SCIENCE CHINA Information Sciences



• REVIEW •

February 2023, Vol. 66 121301:1–121301:31 https://doi.org/10.1007/s11432-022-3566-4

Applications and prospects of artificial intelligence in covert satellite communication: a review

Kun LU¹, Heng LIU¹, Liang ZENG^{2*}, Jiayu WANG¹, Zhongshan ZHANG² & Jianping AN²

¹School of Information and Electronics, Beijing Institute of Technology, Beijing 100081, China; ²School of Cyberspace Science and Technology, Beijing Institute of Technology, Beijing 100081, China

Received 9 March 2022/Revised 27 May 2022/Accepted 3 August 2022/Published online 11 January 2023

Abstract Satellite communication has the characteristics of wide coverage and large communication capacity, and is not easily affected by land disasters. It is quite suitable as a supplement to terrestrial communication networks and has been widely used in education, navigation, emergency relief, military, etc. However, due to the openness of the channel of satellite communication systems, satellite communication signals are easily eavesdropped on by eavesdroppers. This greatly threatens the privacy and security of countries and individuals. Covert satellite communication can effectively improve the covertness of satellite communication systems and greatly reduce the probability of detection by eavesdroppers. So, it has attracted more and more attention. In addition, with the development of artificial intelligence (AI), AI has been applied in many technics of covert satellite communication, which has achieved higher reliability and stronger concealment in covert satellite communication systems. The research status of key technics in covert satellite communication is discussed in this study, and the applications of AI in covert satellite communication are shown. Finally, future research directions of covert satellite communication are looked forward to. In the future, covert satellite communication technology will be an indispensable part of satellite communication systems.

Keywords satellite communication, covert communication, secure communication, low probability of detection (LPD), artificial intelligence (AI), machine learning (ML)

Citation Lu K, Liu H, Zeng L, et al. Applications and prospects of artificial intelligence in covert satellite communication: a review. Sci China Inf Sci, 2023, 66(2): 121301, https://doi.org/10.1007/s11432-022-3566-4

1 Introduction

In recent decades, the speed and range of communication are increasingly required [1,2]. Satellite communication has many advantages, such as wide coverage and large communication capacity, and it is not easily affected by land disasters. Compared with the terrestrial communication network, the satellite network has stronger stability and security, so it is quite suitable as a supplement to terrestrial communication networks. Satellite communication has been widely used in education, navigation, emergency relief, military, etc. However, due to the openness of the channel, the security of satellite communication systems is seriously threatened [3, 4]. The signal of satellite communication systems can be received by any equipment covered by the satellite beam which is a huge threat to both personal privacy and national privacy [5]. What's more, the receiver will also sustain interferences, which may reduce the performance of satellite communication systems. Covert satellite communication is a technology that is able to improve the covertness and security of satellite communication. It can reduce the possibility of communication signals being discovered while ensuring communication reliability. In sum, the development of covert satellite communication technology is crucial to the security of satellite communication systems. The scenario of covert satellite communication described above is shown in Figure 1. Covert users convey information to ground stations through satellites, and eavesdroppers are also under the coverage of satellite beams or are close to the covert users. At the same time, covert users are also faced with unintentional or malicious interference. So, covert users need to take some methods to make the

^{*} Corresponding author (email: liang@bit.edu.cn)



 ${\bf Figure \ 1} \quad ({\rm Color \ online}) \ {\rm The \ scenario \ of \ covert \ satellite \ communication}.$

signal evade the detection of eavesdroppers, and ensure the reliability and security of the communication at the same time.

The research on covert communication technology started in the 20th century [6]. In the beginning, researchers mainly used short-frame burst communication [7] and spread spectrum (SS) [8] to improve the covertness of communication. The SS technology mainly includes direct sequence spread spectrum (DSSS) [9, 10] and frequency hopping spread spectrum (FHSS) [11]. It is a kind of signal structuring method by which a signal can be generated with a much wider bandwidth than the original information. This abuse of spectrum resources can be rewarded with both the lower power density in the frequency domain and noise-like waveform in the time domain, thus significantly improving the covertness of the communication system [12]. Later, the satellite overlap communication was proposed based on DSSS technology. In this way, the DSSS signal is added to the general user signal of the satellite to share the same channel [13,14]. Because of the low probability of detection (LPD) of the DSSS signal, overlapping communication will not affect the performance of the general user signal. This method not only enhances the covertness of the DSSS signal but also effectively improves the utilization of spectrum resources [15].

However, with the development of communication reconnaissance technology, a large number of methods have been proposed to effectively detect and estimate the key parameters of periodic short code DSSS signal [16]. These parameters include carrier frequency, spread rate, spread code period, bit rate and modulation mode [17,18]. Therefore, the DSSS technology cannot meet the current requirements of satellite communication for covertness performance. In order to carry out the research on covert communication more scientifically, Wyner [19] proposed the classic prisoner communication model of covert communication in 1975. The model consists of Alice, Bob, and Willie. For successful covert communication behavior. Based on this model, a set of LPD theories had been developed [20], which realized theoretical analysis of the performance of covert communication systems.

Obviously, according to the LPD theory, both reliability and covertness are indispensable in the covert satellite communication system, which are the two we are most concerned about. The methods to improve the performance of covert satellite communication systems are discussed from the aspect of technical application.

First of all, from the process of transmitting, propagating, and receiving, the methods to improve the

performance of covert satellite communication are considered.

The direct way to improve the covertness of communication is to reduce the transmission power and let the eavesdropper cannot detect the signal. However, too low transmit power may lead to a too low signal-to-noise ratio (SNR) and reduce the performance of modulation. So, how to control transmit power [21] is a key technique in covert satellite communication.

In the propagation process of covert satellite communication signals, the large dynamic characteristics between the satellite and the ground [22], especially in the low earth orbit (LEO) satellite communication system, make it difficult for the satellite antenna to target the terrestrial user. When the angle error of the antenna is too large, the user signal will fall in the sidelobe of the antenna beam. So, the transmit power must be increased to ensure the SNR of the receiver is big enough for modulation, which will result in reducing the covertness and introducing unnecessary interference in the space. In order to solve this problem, the phased array antenna technique [23, 24] is applied, which can flexibly adjust the direction diagram of the antenna by controlling the weighted value of radiation elements in the array antenna. By using the phased array antenna technique, the angle error of the antenna is greatly decreased and the reliability and concealment of covert satellite communication systems are effectively improved.

In general, the effective signal power is awfully low at the receiver of covert satellite communication systems. What's more, because of the openness of the covert satellite communication channel, the effective signal received by the receiver will be superimposed on by the intentional and unintentional interference in the surrounding environment. This will make the receiver work under a low signal-to-interference ratio. Although SS techniques that are mostly used in covert satellite communication have certain anti-interference capabilities, too much interference will still have a serious impact on the performance of covert satellite communication systems [25]. Therefore, the anti-interference technique which can reduce the effect of interference is necessary to improve the reliability of covert satellite communication systems.

In the above, the key techniques of covert satellite communication are discussed from the process of signal propagation. Next, we will discuss the key techniques of the covert satellite communication system itself.

Most of the time, the channel state information (CSI) of the covert satellite communication changes rapidly. For different CSI, researchers hope to choose a proper communication system that can improve the effective communication rate as much as possible on the premise of ensuring concealment. Spectrum prediction and adaptive modulation and coding (AMC) can help us achieve this destination. Spectrum prediction [26] can effectively predict the future spectrum occupation, and then guide the transmitter to adjust the frequency and bandwidth in real time. In this way, the surrounding spectrum environment will be effectively used to maximize the covertness. After the CSI is confirmed, AMC will be applied to determine which coding algorithm and modulation mode to choose in time, so that the performance of the covert satellite communication system can be optimized.

It can be seen from the above discussion that LPD communication, transmit power control, phased array antenna, anti-interference, spectrum prediction, and AMC are effective means to improve the performance of covert satellite communication. Figure 2 shows where these key techniques are employed in a typical covert satellite communication system, in which the satellite is used as the relay.

Many achievements of artificial intelligence (AI) have been made in recent years. AI includes machine learning (ML) [27], deep learning (DL) [28], reinforcement learning (RL) [29], etc., and has been widely applied in electrical engineering, communication engineering [30, 31], software engineering, biological engineering, medical engineering [32], and financial fields. AI is usually used to complete tasks such as clustering, classification, optimization, and regression [33]. Therefore, many researchers apply AI to solve some complex problems in their field [34]. The field of communication has received many successful applications based on AI. Wang et al. [35] reviewed the achievements of ML algorithms in wireless networks in the past decades. Du et al. [36] introduced and surveyed some state-of-the-art techniques based on AI/ML and their applications in 6G. AI has also made great progress in the key techniques in covert satellite communication. We are going to introduce the six key technologies including LPD communication, transmit power control, phased array antenna, anti-interference, and spectrum prediction in detail, at the same time the application of AI technology in them will be shown.

In recent years, some studies related to this paper have been published. Fourati et al. [37] proposed some limitations of satellite communications and discussed their proposed and potential AI-based solutions, and the application of AI showed great results in a wide variety of satellite communication aspects. Yue et al. [5] summarized the unique security vulnerabilities of LEO satellite communication systems, including the issues of passive and active eavesdropping attacks, interference scenarios, single event upsets, and



Lu K, et al. Sci China Inf Sci February 2023 Vol. 66 121301:4

Figure 2 (Color online) Application of key technologies in covert satellite communication.

space debris. And they discussed the corresponding active and passive security countermeasures. Wang et al. [3] presented a comprehensive overview of the protected satellite communication systems, and they investigated and compared the critical technologies used in the protected satellite communication systems. In the research of covert communication theory, Yan et al. [20] reviewed the related research on LPD communication, and discussed the optimal signaling strategies for transmitting the messagecarrying signal and artificial noise signal for LPD communication from an information-theoretic point of view. However, these articles are only reviews of satellite security communication, and there is a paucity of literature on a detailed summary of AI applications in covert satellite communication. In order to fill this knowledge gap, this study summarizes covert satellite communication from seven key technical points, introduces the application of AI technology, and finally looks forward to the future development trend.

The rest of this paper is organized as follows. Section 1 introduces the background and key techniques of covert satellite communication systems, and gives an overview of the development of AI. Section 2 talks about some AI techniques used in covert satellite communication. Section 3 discusses six key techniques including LPD communication, transmit power control, phased array antenna, anti-interference, spectrum prediction, and AMC, and introduces their development statuses based on AI. Section 4 looks forward to the future development of AI in covert satellite communication. Section 5 summarizes the full text. The acronyms used in this paper can be found in Table A1.

2 Related artificial intelligence methods

AI was first proposed at the Dartmouth Institute in 1956. It is the technology that presents human intelligence through ordinary computer programs. This technology aims to make machines smarter by following some rules or by facilitating guided learning. With the development of computer science, the theoretical methods of AI are gradually enriched. Researchers are beginning to apply AI methods to various fields [1, 38, 39]. Four AI methods will be introduced: convolutional neural network (CNN), long short-term memory (LSTM), Q-learning, and generative adversarial network (GAN). These are four classical AI techniques, and their research has obtained rich achievements and wide applications.

2.1 Convolutional neural network

CNN is a kind of feedforward neural network with convolution calculation and deep structure, and it is one of the representative algorithms of DL. The structure of CNN is also divided into the input layer, hidden



Figure 3 (Color online) The structure diagram of CNN.

layer, and output layer, which is the same as other DL networks. The convolutional layer, pooling layer, and fully connected layer are included commonly in the hidden layer according to different algorithms, while the convolutional layer and pooling layer are unique to CNN. Its structure diagram is shown in Figure 3. In addition to the differences in the structure of each layer itself, with the development of modern algorithms, people have invented complex network structures such as the inception block, the residual block, the dense block, and the attention mechanism. The structures are used to solve various problems encountered by deep CNN models.

The good local connectivity of the CNN algorithm can extract local features of objects and effectively retain useful information. And the feature of weight sharing greatly reduces the number of parameters and thus reduces the difficulty of training. More and more researchers have combined the CNN algorithm with satellite communication technology in recent years. For example, CNN is used to identify and classify interference in satellite communication scenarios [40, 41], or to extract and separate effective signals from interference [42]. Automatically identify the modulation format in terms of satellite modulation technology so that the receiver can perform adaptive demodulation [43, 44]. And also the construction and synthesis of phased array antennas for satellite communication transmitting equipment [45], the prediction of the future spectrum in communication environments [46], the control of the transmit power strategy in power control to maximize spectral efficiency (SE) [47, 48] and so on. There will also be more in-depth research directions and broader application fields in the field of satellite covert communication.

2.2 Long short term memory

A recurrent neural network (RNN) is an algorithm used to process sequential data. Compared to a normal neural network, it can process changing data. However, RNN can only memorize short time-series data. As the number of data and the time interval increase, RNN will lose the important information previously input, causing the gradient to disappear and leading to the failure of the prediction model [49].

LSTM is a special RNN proposed by Hochreiter et al. [50] in 1997. LSTM network is suitable for processing and predicting important events with relatively long intervals and delays in time series. Compared with ordinary RNN, LSTM can perform better in longer sequences. Therefore, LSTM can solve the problems of gradient disappearance and gradient explosion in the long sequence training process of the RNN algorithm. The LSTM network processes the input time series by transferring the transfer function, and Figure 4 shows the structure of a memory cell in the LSTM network. LSTM has three gates in one cell, which are called input gate, forget gate, and output gate, respectively. The input gate determines whether the hidden layer information is updated. The forget gate determines whether the updated information contains information from the previous moment. The output gate determines which part of the updated information to output. When a message enters the LSTM network, it can be determined whether it is useful or not according to the rules. Only the information conforming to algorithm authentication will be left behind, and the inconsistent information will be forgotten through the oblivion gate.

In recent years, LSTM has been widely applied in the field of satellite communication. Gunn et al. [51] used the LSTM algorithm to predict errors in satellite communication monitoring, and the results show that this method has better performance than the simple threshold model. Lee et al. [52] applied LSTM to the synchronization of frequency-hopping signals in a tactical satellite communication system. The



Figure 4 (Color online) The structure of the memory cell.



Figure 5 (Color online) The flow chart of Q-learning.

LSTM network is used to learn the trend of energy signals for fine acquisition. With the development of covert satellite communication, some scholars have combined the LSTM network with covert satellite communication technology and achieved a lot of research results [53–56].

2.3 Q-learning

RL describes the process by which an agent interacts with the environment. Maximize the returns or achieve specific goals through certain learning strategies. Q-learning is a value-based algorithm in RL algorithms, which is a decision-making process. Q is Q(s, a), which is the expectation of the reward that the action a can obtain at a certain moment in the state. The environment will feedback corresponding rewards according to the actions of the agent. So the main idea of the algorithm is to build state and action into a Q-table. Each row of the Q-table represents each state, each column represents each action, and the Q value (Quality) is stored in the middle. According to the Q value, the action that can obtain the maximum benefit will be selected. The algorithm continuously updates the Q value during the process and the flow chart of this algorithm is shown in Figure 5.

Q-learning uses the time difference method, and integrates Monte Carlo and dynamic programming. It has the ability to perform off-policy learning and the optimal strategy for the Markov process can be solved by the bellman equation. It is a typical model-independent algorithm that can perform operations without establishing an environment model, making it highly versatile. And it is not limited to episode tasks, requiring fewer parameters. Any task that can build a Q-table can use the Q-learning method for modeling, which makes Q-learning widely used in various strategies and games.

Tsuchida et al. [57] applied Q-learning to the problem of power distribution in satellite-ground communication of LEO satellites, and the results show that the proposed method can prolong the battery life of LEO satellites. To solve the high handoff dropping probability caused by the long transmission delay of satellite links, Xiong et al. [58] proposed a novel handoff decision strategy based on the predictive





Figure 6 (Color online) The basic model structure of GAN.

received signal strength (RSS) and Q-learning algorithm. In covert satellite communication technology, Q-learning also plays an important role in some aspects. For example, the problem of channel selection in the complex environment of satellite communication systems [59], the optimal game problem between the interferer and the interfered [60,61], etc., effectively reduce the interference in space impact on effective communication. Or under the needs of complex communication networks, use Q-learning to achieve optimal real-time power allocation [62]. It can be seen that the combination of RL up to Q-learning and satellite communication has great research potential.

2.4 Generative adversarial network

ML can be divided into supervised learning and unsupervised learning. Although supervised learning has achieved good performance, it needs to rely on known labeled data, and the algorithm cost is high. In order to reduce the cost of learning, unsupervised learning has received more and more attention, and generative models are one of the most promising techniques [63]. Early generative models, such as restricted Boltzmann machines, deep belief networks, and deep Boltzmann machines, have poor performance in generalization. In response to this problem, in 2014, Goodfellow et al. [64] introduced the concept of GAN. By 2016, researchers began to discover the great potential of GAN, and GAN broke through the bottleneck that previously limited the development of DL [65].

The ultimate goal of GAN is to understand the appearance of real data, estimate the distribution or density of real data, and finally generate new data based on the learned knowledge. GAN is inspired by the two-person zero-sum game in game theory, in which the sum of the interests of both parties is 0 or remains the same, so one party gains while the other party must lose. In this way, the two neural networks are constantly learning and optimizing in the process of playing against each other, and finally, the optimal model will be achieved.

The basic model structure of GAN is shown in Figure 6. It mainly consists of two important parts, that is, the discriminator D and the generator G. Among them, the generator takes random noise as input and learns the distribution of real samples, so as to be able to generate sample data close to the real. The discriminator takes real data or the generated data of the generator as input and tries to determine whether the input data is real data or generated data. These two parts iteratively optimize their performance through continuous adversarial training. When the data generated by the generator G are as close to the real data as possible and the probability of the discriminator's output is close to 1/2 each time, the GAN reaches the Nash equilibrium.

GAN has many advantages. It can learn the true distribution of samples and achieve strong predictive ability according to the distribution of samples. Moreover, GAN is very robust in generating samples, which is unattainable in many ML models. The most direct application of GAN is data modeling, which generates samples that are consistent with the real data distribution, and it is widely used in image generation, image classification, target detection, etc. [66]. Due to the powerful prediction generation ability of GAN, it has also received extensive attention in the field of covert satellite communication. Guan et al. [67] proposed a satellite downlink signal detection method based on GAN. Davaslioglu et al. [68] and Lin et al. [69] used GANs to enhance sparse historical data for more stable and robust spectral predictions. Liao et al. [70] proposed a GAN-based power allocation algorithm to solve the power allocation problem of a cooperative cognitive covert communication system.



Figure 7 (Color online) Key techniques for covert satellite communication based on AI.

3 Applications of AI in covert satellite communication

In this section, the applications of AI in six key techniques for covert satellite communication are discussed, including LPD communication, transmit power control, phased array antenna, anti-interference, spectrum prediction, and adaptive modulation and coding. Figure 7 shows these six key techniques in covert satellite communication. Next, they will be discussed separately.

3.1 Low probability of detection communication

3.1.1 Definition & limitations

Because of the openness of the wireless communication system, the transmitted information is easy to be intercepted or attacked maliciously. Therefore, information security has always been the hotspot of research in wireless communication. LPD communication is a technique that ensures wireless security at the physical layer through covert signals. It can greatly reduce the detection of communication signals by the eavesdropper and achieve the purpose of concealing signals by coding, modulating, designing waveform, and other methods [71].

Figure 8 describes the scenario of a covert satellite communication system, which consists of a satellite, covert users, stations, and eavesdroppers. Covert satellite communication is expected to reduce the probability of communication being detected by the eavesdropper on the premise of ensuring the stable transmission of information between the station and the covert user. Especially in some military communication scenarios, communication signals are extremely undesirable to be detected by the enemy. So higher requirements for information security are put forward. LPD technique can fundamentally solve the malicious behavior of signals, so the study of LPD has become an important technical support for the realization of covert satellite communication. The problems and solutions of LPD communication are described as follows.

The basic model of the LPD problem is shown in Figure 9. This classic prisoner communication model was first proposed by Wyner in 1975 [19], which consists of three basic components: Alice (the signal sender), Bob (the legal receiver), and Willie (the illegal monitor). Prisoner Alice needs to transmit critical information to Bob. And in the meantime, their communication process cannot be detected by Willie. Willie does not need to pay attention to the contents of communication, but only to whether communication occurs. Therefore, the goal of LPD is to ensure the reliable transmission of information between two users while ensuring that the probability of the information being detected by the monitoring party is low enough [20].

Traditional LPD communication mainly relies on SS communication technologies, including DSSS [72] and FHSS [73–77]. The schematic diagrams of the two techniques are shown in Figure 10. SS communi-



Figure 8 (Color online) LPD satellite communication model.



Figure 9 (Color online) The basic model of the LPD communication.

cation keeps the signal power unchanged and reduces the signal power spectrum density to below noise by expanding the signal spectrum bandwidth, so that it cannot be noticed. Therefore, SS communication can realize covert communication in a low SNR environment. However, the SS technique is mostly used in engineering practice, and there is no theoretical limit on the system capacity of this method. Therefore, the security performance of SS communication cannot be quantified. With the development of wireless techniques, many detection methods for SS communication have been proposed [78, 79]. The current research on LPD mainly focuses on LPD information theory, implementation methods, constant rate covert communication, and so on.

3.1.2 Main challenges and solutions

The information theory of LPD communication has achieved a lot of progress. In 2013, Bash et al. [80] studied LPD communication from the perspective of information theory and proposed the concept of covert communication. The results showed that the communication capabilities that can be reliably transmitted over an additive white Gaussian noise (AWGN) channel are limited by the square root law (SRL) for LPD communication. SRL says that covert communication can transmit $O\sqrt{n}$ bits information for blocks of length N. The communication rate decreases with the increase of the code length,



Lu K, et al. Sci China Inf Sci February 2023 Vol. 66 121301:10

Figure 10 (Color online) Spread spectrum techniques. (a) DSSS; (b) FHSS.

and the traditional Shannon capacity tends to zero when the transmission code length tends to be infinite. Subsequent studies showed that SRL is suitable for LPD problems of various channels. Abdelaziz et al. [81] proved that SRL is suitable for the MIMO Gaussian channel and the number of bits that can be transmitted covertly has an exponential relationship with the number of transmitting antennas. Wang et al. [82] considered the LPD problem over a discrete memoryless channel (DMC), and gave the maximum information and block length that can be transmitted under SRL conditions. Tahmasbi et al. [83] clarified how the choice of imperceptibility measure affects the information theory constraints of covert communication over DMC. Bloch [84] developed an encoding scheme for a binary symmetric channel (BSC) based on the principle of channel resolvability. Che et al. [85] proved that if the legal channel environment is superior to the monitoring channel, the effect of covert communication can be achieved even if both parties communicate by using public codes instead of common keys, which is also a communication model based on BSC.

Based on the support of theoretical research, a lot of research has been made on the implementation of the LPD technique. Bloch et al. [86] used a random m-element external code and an m-order pulse position modulation (PPM) internal code cascade, which implements the information-theoretical limits of covert communication. Kadampot et al. [87] proposed a combination of PPM and multi-level coding, which is a low-complexity coding scheme. Li et al. [88] made a joint optimization of the modulation order L and transmitted power P of PPM in covert communication. Yan et al. [20] studied AWGN channel covert communication with short delay constraints and proved that effective throughput of the system can be improved if all available channels were utilized by transmitters.

A lot of research achievements have also been achieved in the constant rate of covert communication. This method makes use of the uncertainty of the communication system or human interference to produce detection errors, which reduces the monitoring ability of the monitor but does not affect the legitimate communication performance too much.

There are many uncertainty factors in communication systems, such as noise uncertainty, communication transmission time uncertainty, and channel uncertainty. Considering the communication with noise uncertainty distribution, He et al. [89] proposed the maximum realizable communication rates under bounded or unbounded noise uncertainty models. Ta et al. [90] analyzed the combined effects of channel uncertainty and noise uncertainty on the average detection error probability and throughput of the monitor. It is proved that channel fading is very important for covert communication. Bash et al. [91] proposed that constant rate covert communication can be achieved by selecting a random time slot communication known by both communicators and unknown to the monitor, but a sufficient number of alternative time slots are required.

Some scholars considered adding artificial noise to the original LPD communication model. This approach weakens the administrator's ability to detect traffic by confusing eavesdroppers. Sobers et

al. [92,93] proposed that the LPD communication model consists of transmitters, receivers, and friendly jammers, and the jammers produce artificial noise. The jamming activity is independent of the legitimate communication process so the monitor cannot detect the transmission by listening to the jammer. Shahzad et al. [94] studied the duplex wireless communication system in fading channel, which generated artificial noise of different powers, resulting in the uncertainty of the statistical data of the received signal by the monitor. At the same time, the best choices of power range of artificial noise and the best transmission probability of covert information were given. Yang et al. [95] achieved covert communication by using a full-duplex multi-antenna receiver with random transmit power, and they demonstrated that increasing the number of antennas increases the ability to perform covert communications. He et al. [96] studied the effects of density and transmit power of concurrent interference sources on covert throughput. It is proved that when the interference is small enough and is equivalent to the receiver noise, the throughput of the system increases with the increase of the density and transmission power of the concurrent interference source.

3.1.3 AI-based solutions

With the advent of AI, some researchers began to consider applying ML to LPD communication scenarios. Yang et al. [97] constructed a burst communication model with low intercept probability based on conditional maximum entropy. In this model, hybrid chaotic particle swarm optimization (HCPSO) is used to optimize the objective function to obtain the conditional maximum entropy model, which makes the covert communication performance better. Shakeel [98] proposed using ML to generate synchronized featureless signals, which has better LPD performance than the DSSS system. Zhou et al. [99] proposed a low probability of intercept (LPI) communication system based on ML, which has low interception accuracy. The combination of ML and LPI/LPD communication is a very novel direction. Few research achievements have been made so far, and we expect there will be more important progress in the future.

3.2 Transmit power control

3.2.1 Definition & limitations

Transmit power control refers to the technology that timely adjusts the transmit power of the transmitter according to the functional requirements of the communication system. In a communication system, if the transmit power is too low, the receiver cannot demodulate correctly, which will reduce the system performance. On the contrary, if the transmit power is too high, it will waste the power and spectrum resources of the system, and may cause interference to other communication devices in the same network [100,101]. It can be seen that power control is extremely important for the performance improvement of communication systems, and this technique has been widely used in mobile communication networks [102–104], Internet of Things (IoT) [105], and satellite communications [106].

In covert satellite communication, if the warden (Willie) tries to identify the presence of effective communication by standard methods (such as energy detection in each spectrum band), we have a motivation to reduce the transmit power in each band to the minimum value, to hide the behavior of communication [107]. As known from the noisy channel coding theorem [108], even at extremely low signal-to-noise ratios, a reliable communication channel can be established at the expense of reducing the transmission rate to be lower than the channel capacity. Reasonable control of the transmit power can not only enhance the throughput and robustness of the covert satellite communication but also reduce the probability of detection of the warden. In sum, transmit power control is crucial for improving the reliability and covertness of covert satellite communication.

3.2.2 Main challenges and solutions

At present, the research on transmit power control in satellite communication can be divided into the following three categories.

One is to maximize the rate or throughput of the network by allocating and controlling the power of each node in the network [106,109,110]. Zou et al. [109] studied the optimization of power control in cognitive satellite-terrestrial networks under hybrid interweave-underlay to maximize network throughput. Based on the cognitive model of CSMA/CA (carrier sense multiple access with collision avoid), the strong interference of secondary users to primary users is limited from the two dimensions of space and time. In order to ensure the communication QoS of terrestrial primary users who have both delay-sensitive and

delay-insensitive communication service requirements, Gao et al. [110] proposed a satellite uplink power control algorithm based on the alternating direction method of multipliers (ADMM). Wang et al. [106] studied a distributed spectrum access and power control scheme based on ML for multi-beam satellites in a non-cooperative game cognitive mode to maximize the achievable rate of the communication network.

The second category is to improve the robustness of the network under non-ideal conditions such as large dynamics, interference, and jammer, to minimize the probability of interruption [111–113]. Vassaki et al. [111] modeled with the constraints of the quality of service (QoS) requirements of the terrestrial network and the interruption probability of satellite communication. Then, based on the constraints, they proposed an optimal power control scheme for the terrestrial network as a secondary user (SU). Considering the influence of inaccurate CSI estimation for satellite communication links and terrestrial interference links, Shi et al. [112] proposed a power control scheme that maximizes outage capacity. Based on the dynamic characteristics of LEO satellites, Hu et al. [113] proposed a mathematical model for satellite-to-earth cognitive scenarios based on LEO constellations and terrestrial networks. They proposed a power control scheme to maximize delay-limited capacity and minimize interruption probability.

The third category is to minimize the delay of the control network under the condition of long distance between satellites and the terrestrial user [114, 115]. Aiming at the problem of interference management in the terrestrial primary network, Shi et al. [114] proposed two power control schemes suitable for realtime communication scenarios. They adopted the average interference power constraint and the peak interference power constraint to maximize the system delay-limited capacity. To solve the problem of unsatisfactory spectrum sensing and the delay caused by the mobility between the satellite and the terrestrial user, Wang et al. [115] proposed a distributed joint resource optimization scheme, which minimizes end-to-end data transmission delay for the SU.

For the problem of transmit power control in covert communication systems, a lot of research has been done, and the main problem is to maximize the sum of the effective communication rates of the network under the constraints of covertness. Different from the traditional communication network, the goal of covert communication is not only to improve the communication performance but also to reduce the detection performance of the warden. So, the covert communication system is unique in resource optimization. Considering the communication over parallel Gaussian channels under an LPD constraint, Letzepis et al. [116] analyzed the optimal power allocation strategy that can maximize the achievable number of transmitted information bits per codeword under constraints on the receiver's decoding error probability and the adversary's detection probability. Dvorkind [107] considered the problem of spectral power allocation under the condition of ensuring that the warden's SNR is below a certain threshold. Forouzesh et al. [117] and Hu et al. [118] studied covert communication transmit power control in relay network scenarios. Zheng et al. [119] considered the multi-antenna and multi-jammer scenarios and determined the optimal detection threshold from the perspective of wardens. They obtained the maximum covert throughput by jointly optimizing the transmit power and transmission rate under this condition.

Recently, some scholars have studied the channel inversion power control (CIPC) technology, which can effectively improve the covertness of communication systems [105, 120–122]. Hu et al. [120] studied CIPC-based covert communication under Rayleigh fading. Specifically, the transmitter varies the power and phase of transmitted signals as per the channel to the receiver, such that the receiver can decode these signals without knowing the CSI. Then, they applied it to IoT [105]. Ma et al. [121] studied covert communication with finite blocklength under the CIPC scheme, which allows the transmitter to communicate with the receiver without transmitting a pilot signal and can improve the effective covert throughput. Wang et al. [122] proposed a scheme based on CIPC, which instructs public users as a random transmit power jammer to interfere with the warden's detection.

The classification of the above research on power control is shown in Figure 11.

3.2.3 AI-based solutions

With the development of DL, more and more people began to research applying the deep neural network (DNN) to optimize power control in wireless communication. According to the universal approximation theorem of DNN [123], any continuous function can be approximated by DNN. So, DNN tried to approximate the optimal transmit power control parameters that could not be reached before. What's more, it can also reduce the calculating time. Sun et al. [124] used DNN to approximate the near-optimal transmit power control of the weighted minimum mean square error (WMMSE), resulting in a reduction in computation time. Lee et al. [47] proposed the first CNN-based transmission power control framework,



Figure 11 (Color online) Application classification of transmit power control technologies.

deep power control (DPC), in which the transmit power control strategy to maximize the SE or energy efficiency (EE) can be achieved through CNN learning. The proposed DPC form can achieve almost the same or even higher SE and EE compared with the traditional power control scheme, and a lower computation time is also achieved. Then they used DNN to design a multi-channel underlay cognitive radio network (CRN) power allocation scheme [125]. They also put QoS into constraints and proposed DNN-based transmit power control strategies [126, 127]. Lee et al. [48] used CNN to investigate the joint optimization of transmit power and time switching ratio. As a result, the EE of a wireless-powered network with co-channel interference was maximized.

Although DNN-based power control methods have been extensively studied, the above methods are almost all proposed under the assumption that the global CSI is well known. However, it is extremely difficult and unrealistic to obtain accurate global CSI in time in practice. In response to this problem, Lee et al. [128] proposed a distributed DNN-based underlay CRN transmit power control scheme. In this scheme, each SU can determine a near-optimal transmit power policy based only on its local CSI, and compared to centralized DNN-based schemes this one can greatly reduce operational overhead while maintaining high performance.

To meet the demands of more and more complex communication networks, deep reinforcement learning (DRL) began to be introduced into power control algorithms. Considering the practical scenarios where the system model is inaccurate and the CSI delay cannot be ignored, Nasir et al. [62] proposed a distributed dynamic power allocation scheme based on model-free DRL, which can achieve near-optimal real-time power allocation based on delayed CSI measurement data. To solve the high-dimensional problem in transmit power control, Zhang et al. [129] proposed a power control method by asynchronous advantage actor critic (A3C) based on RL and DNN, which achieved great performance in power efficiency and network convergence. For overcoming the difficulties caused by the random variation of the RSS collected by the sensor, Li et al. [130] used DRL to assist the transmit power control and ensure the QoS of users in the network. Tan et al. [131] developed a distributed DRL-based power allocation scheme in an overlay device-to-device network scenario, which achieves better scalability compared to FP-based schemes.

For AI-based methods to improve the covertness of communication systems, Liao et al. [70] studied the power allocation problem of a cooperative cognitive covert communication system with a relay station, and proposed a power allocation algorithm based on a GAN. This is the first time that GAN has been used as a transmit power control method to improve covertness. The current AI technology is mostly applied to the optimization of the maximum transmission rate and throughput of the communication network, and further research is needed to improve the covertness.

3.3 Phased array antenna

3.3.1 Definition & limitations

The phased array antenna has multiple radiating elements arranged and combined. By controlling the weighted value of the radiating elements in the array antenna, the direction of the maximum value of the

pattern can be changed to achieve the purpose of beam scanning [132, 133].

In covert satellite communication, how to hide the communication process is very important. The most common method is to reduce the power of the transmitted signal so that the signal drowns in the noise. In order to improve the receiving reliability, it is necessary to design a receiver system with strong anti-interference ability. The phased array antenna can accomplish this task well. In current satellite communication systems, the high speed and complex motion of the ground terminal bring great difficulties to the efficient and stable communication of the system. Beamforming technology forms an ideal signal in the specified direction, improving the efficiency of spectrum utilization. Therefore, the phased array antenna system has automatic tracking technology and can realize fast acquisition through a smart antenna. It provides important technical support for the satellite communication system [134]. When the phased array antenna is transmitting, it can generate beams with high gain and low sidelobe level by its flexible beamforming technology, so that the satellite signal is not easy to be intercepted by the enemy. For the receiver, the zero point of the antenna pattern can be adaptively aligned with the direction of interference to ensure that the gain of the useful signal is almost unaffected. Therefore, the interference and error of the received signal are controlled to the minimum, so that the satellite transponder can still work normally in the interference environment [135].

3.3.2 Main challenges and solutions

A good antenna array arrangement is the basis of a phased array antenna system. In order to adapt to different satellite communication scenarios, the phased array has unique characteristics in hardware design. Kapusuz et al. [136] proposed a design scheme of an 8×8 phased array antenna for Ku-band mobile satellite communication. The array had the characteristics of low profile, high gain, and wide-angle scanning so that the satellite communication terminal would have a relatively compact size. Simulation results showed that the phased array antenna can scan 50 degrees in both axes without generating any grating lobes. Rao et al. [137] proposed a unique phased array concept of a planar array antenna, which can generate multiple independent beams to communicate with multiple geostationary satellites simultaneously. It has the capability of omnidirectional electronic beam scanning which can reduce the cost and complexity of phased array antennas. Zhang et al. [138] proposed an active phased array antenna (APAA) for Ka-band broadband mobile satellite communications. They discussed a highly integrated antenna module that combines the antenna, radio frequency (RF) circuitry, and cooling system. The results showed that the axial ratio of a single element is about 4 dB on the boresight, the cross-polarized lobe that appears is only 7 dB below the main beam level, and in the steering beam, the axial ratio is less than 2.0 dB.

In order to meet the requirements of signal transmission, the phased array antenna as a transmitting device needs to design a signal transmission algorithm and control beam configuration to cope with the rapid growth of satellite communication throughput. Kumar et al. [139] used the spherical phased array antenna for systems with small beam directivity changes. By combining different output amplifiers to accommodate information transmission at different data rates, there is enough equivalent isotropic radiated power (EIRP) to meet link requirements. Moon et al. [140] proposed a beamforming technology based on enhanced beam steering, which can support enhanced mode under any weather conditions to meet the high-power signal of communication. To achieve effective satellite tracking, Raj et al. [141] utilized a hexagonal 7×7 antenna array consisting of built-in phase shifters. A structured beam in the desired direction is obtained by eliminating the beam in the unwanted direction with rear lobe suppression.

The phase and amplitude imbalance of the radiation antenna elements is caused by the mode difference of the embedded elements, the gain and phase offset of the RF electronic devices, and the temperature change of the aperture. For the normal use of phased array antennas, most systems need to be calibrated during operation. Calibration can achieve accurate beam pointing and reduce the error of the hardware system. Lin et al. [142] proposed an online phased array antenna calibration method based on partial parallel interference cancellation multi-user detection, which would not interrupt normal communication. This method sends the modulation low power calibration signal and the normal communication signal at the same time, the existence of the calibration signal can maintain the reliability of the normal communication. The simulation results showed that the calibration accuracy of the calibration algorithm can achieve near-linear multiuser (LMU) calibration accuracy with low computational complexity. Ghaffarian et al. [143] used a planar near-field probe set up in an APAA environment for external calibration. They validated a phased array antenna consisting of 1024 radiating elements over a wide scan angle of 18–21 GHz (0° –70°).

3.3.3 AI-based solutions

With the rise of AI, research on the combination of AI and phased array antennas has also attracted widespread attention in the industry. For the three scenarios mentioned above, the addition of AI technology has led to technological innovation to a certain extent.

For the construction and synthesis of transmitting active phased array antennas, Mishchenko et al. [144] created an antenna algorithm that can combine multiple continuous stages. A neural network structure consisting of a classified neural network and several approximate neural networks is constructed to solve the structural synthesis problem of the transmitting active phased antenna array. Elbir et al. [45] proposed an antenna element selection method based on CNN to find the optimal subarray. This method does not need to compute analog and baseband beamformers, but only needs to feed the estimated channel matrix back to the network. Considering the time-varying of channel and user parameters during simulation, training the network with a large amount of training data (240000 input samples), and a noisy channel matrix, the classification accuracy quickly reached 100%.

For the design of the beamforming algorithm, Lovato et al. [145] combined AI with electromagnetics. Given the required two-dimensional radiation pattern, the neural network is used to calculate the phase of the antenna array in complex radiation mode. Simulation verified that it can accurately calculate the phase value of array beamforming. Fournier et al. [146] implemented a neural network-controlled beamforming network on field-programmable gate array (FPGA) to achieve a more flexible, faster, and more reliable control method. Hameed et al. [147] used the DNN algorithm to find the optimal solution for the energy beamforming vector in multi-antenna non-communication networks. Offline training can be more efficient and less complex. The simulation results proved that the DNN scheme has lower computational complexity and time complexity than the traditional sequential parametric convex approximation method.

For phased array antenna calibration, Sarayloo et al. [148] proposed a calibration method based on amplitude measurement and DL for phased array calibration. Under different SNR conditions, the phase calibration coefficients of each RF path of the phased array receiver are estimated by convolution and multilayer perceptron neural network. Thus, the errors affecting the phased array performance can be reduced with higher speed and accuracy. Southall et al. [149] proposed a neural network algorithm for digital beamforming in the case of uncalibrated antennas or poor performance due to unknown degradation. The proposed adaptive radial basis function artificial neural network (ANN) architecture can learn single-source direction finding functions for octa-element X-band arrays with multiple unknown failures and degradations.

3.4 Anti-interference

3.4.1 Definition & limitations

Interference signals can be roughly divided into man-made Interference and non-man-made Interference, both of which will reduce the communication quality to a certain extent. Man-made interference is divided into intentional and unintentional. Generally speaking, intentional interference is a malicious signal sent by attackers to jam with the communication process. Unintentional interference is intrasystem interference or inter-system interference which is caused by insufficient system design or improper frequency management between systems [5].

The scattering of electromagnetic waves by the atmosphere, the reflection and refraction of electromagnetic waves by the ionosphere, the reflection of electromagnetic waves by surface objects such as mountains and buildings, and the influence of bad weather will all cause multipath propagation, resulting in signal attenuation and phase shift. Therefore, non-human interference mainly includes multi-path interference, atmospheric scintillation or fading, which are mostly caused by nature. The schematic diagram of interference classification is shown in Figure 12.

The covert satellite communication environment has a large dynamic change, and there are many interference factors, such as common multipath, intra-system interference, inter-system interference, and false signals emitted by some devices, etc. [150]. Generally, the covert satellite communication system has higher requirements on the reliability, accuracy, and security of the transmitted information, so the



Figure 12 (Color online) The schematic diagram of interference classification.

system needs to have an excellent anti-interference performance to ensure the success of the covert satellite communication process. However, a considerable part of covert satellite communication systems is built on the background of the SS communication technique, which consumes frequency band resources in exchange for better anti-interference and anti-noise performance. Therefore, the spread-spectrum system has a certain capability to be immune from interference, as shown in Figure 13(a). But when the strength of the jamming source exceeds the anti-interference ability of the system, the system performance will be seriously degraded, and even lead to the paralysis of the system [151]. The existence of interference is unavoidable, so it is very necessary to adopt some technologies to protect the covert satellite communication system in the interference environment.

3.4.2 Main challenges and solutions

At present, for the detection and suppression of interference, there are many anti-interference processing technologies widely used in the scenario of covert satellite communication, which can be divided into the time domain, transform domain, time-frequency domain, time-space domain, and so on.

Among the time-domain methods, adaptive filtering techniques using finite impulse response (FIR) and infinite impulse response (IIR) filters are superior filtering methods to eliminate interference and have attracted extensive attention. Figure 13(b) vividly describes the signal-changing process under the time-domain adaptive filtering. An adaptive IIR notch filter requires considerably fewer filter coefficients compared with FIR for the same notch bandwidth while demonstrating quite similar performance to that of the FIR [151]. For continuous wave interference (CWI), Chien [152] and Lv et al. [151] used an adaptive notch filter (ANF) to estimate the power of the detected CWI by exploiting the statistics and internal states associated with the IIR filter and the CWI signal in the time domain was effectively detected and mitigated. Mao et al. [153] discussed the application of the adaptive all-pass-based notch filter (ANFA) in the improved adaptive Gauss-Newton (MAGN) algorithm for four kinds of continuous-wave interference signals. ANFA has a higher SNR and lowers residual interference level, which is superior to the traditional adaptive filter.

Among the transform domain methods, techniques used are usually based on the spectral estimation of the incident signal to track the rapidly changing interference. Figure 13(c) is a schematic diagram of the process of real-time interference elimination in the transform domain by using an adaptive threshold. Wang et al. [154] designed a partial coefficient update least mean square adaptive algorithm with local



Figure 13 (Color online) Three ways for anti-jamming by filtering. (a) Spread spectrum receiver; (b) time-domain adaptive filtering; (c) frequency-domain adaptive filtering.

coefficients update to suppress narrowband interference, which could adaptively update N groups of interference blocks in the frequency domain. Zhang et al. [155] proposed an interference suppression method combining wavelet transform and frequency shift for the CWI signal that had a great influence on the satellite communication system. It could not only effectively suppress narrowband interference, but also reduce noise and reduce the time of a single stop of acquisition.

Time-frequency domain techniques allow people to observe received signals in a joint domain. Interfering signals are usually concentrated in a limited area of the two-dimensional time-frequency plane, while noise is distributed over the entire plane [150]. Djukanovic et al. [156] used local polynomial Fourier transform to eliminate single-component non-stationary interference. A binary mask was used to remove interference through a time-varying filtering process and modified the receiver to maximize the output SNR. Savasta et al. [150] used a synthesis technique based on orthogonal Gabor expansion. In this method, the interference signal in the time domain is recovered from the Gabor coefficient of the input signal and then subtracted from the input signal to reduce the influence of interference on the required signal in the time-frequency domain.

Among the space domain methods, antenna arrays are usually used to suppress broadband and narrowband interference, which adds a significant anti-interference capability to satellite communication systems. Amin et al. [157] discussed the application of sparse arrays and sparse signal processing for interference suppression and direction of arrival estimation. For multipath interference design, Heng et al. [158] proposed the high mask angle antenna to suppress the interference and signal from the low elevation Angle for multi-path interference design. The optimal mask angles are obtained by different constellation settings, accuracy dilution of precision indices, and range accuracy assumptions. Simulation results showed that the proposed method is beneficial to reduce multipath interference for multi-constellation global navigation satellite system (GNSS) users. Space-time adaptive processing (STAP) combines spatial and temporal information to provide joint spatiotemporal processing to suppress multipath and narrowband and wideband interference. Zhao et al. [159] proposed a scheme to use the STAP algorithm to improve the anti-jamming performance of the antenna array. It was proved by simulation that STAP is superior to spatial adaptive processing in the case of multiple jamming signals.

In addition, we summarize three methods of interference suppression in the communication process in Figure 13. Figure 13 shows three ways of eliminating interference: time-domain adaptive filtering, frequency-domain adaptive filtering, and SS receiver. The signal condition after anti-interference is given, which proves that the bad influence of interference can be removed under the condition of protecting the original signal information as much as possible.

3.4.3 AI-based solutions

The vigorous development of AI has provided many new ideas and methods for the identification and classification and the suppression and elimination of interference. First, AI can assist in identifying the type or characteristics of interference. Second, AI can be applied to the game selection between the interferer and the interfered in a dynamic environment. Third, AI can efficiently adjust the adaptive parameters required by the system.

For the identification and classification of interference, accurate and reliable identification of the type of interference is a necessary prerequisite for the use of targeted anti-interference methods. Refs. [40] and [160] both proposed methods for classifying interference types in satellite communication systems. Ferre et al. [40] proposed an efficient jammer classifier based on support vector machine (SVM) and CNN. First, the spectrogram is calculated from the intermediate frequency (IF) sample, and the jammer classification problem is transformed into a black and white image classification problem. When an ML algorithm is applied to this problem, the classification accuracy can reach more than 90%. Xu et al. [160] proposed a jamming recognition algorithm based on the DNN, which extracts a set of low-complexity and high-resolution jamming features in the time domain, frequency domain, and transform domain, and constructs the disturbance classifier based on the decision tree and DNN, respectively. The simulation comparison results showed that when the jamming SNR is 0 dB, the recognition rate of the typically suppressed jamming by the DNN-based classifier can reach 99%. When the jamming SNR is 10 dB, the recognition rate of the DNN-based classifier for all 10 kinds of composite jamming can reach more than 85%.

In the complex electromagnetic dynamic environment of the satellite communication system, the channel selection problem and the optimization game between the interferer and the interfered are the keys to maintaining the performance of the communication system. Yao et al. [59] defined the channel selection problem of anti-jamming defense as Stackelberg game, and based on stochastic learning theory, established a hierarchical learning framework and proposed a hierarchical and learning-based channel selection algorithm. Erpek et al. [161] designed an intelligent jamming attack system and gave a corresponding defense scheme, where the transmitter senses the channel and applies a pre-trained ML algorithm to detect the idle channel of transmission for data transmission. Noori et al. [60] studied a stochastic game method between jamming and anti-jamming in jamming channels. Combining game theory with RL, Han et al. [61] proposed a distributed dynamic anti-jamming scheme that reduced energy consumption in jammed environments. Lu et al. [162] proposed a relay technique against intelligent jamming, which combines RL with a function approximation approach named tile coding. Based on game theory, the mobility and relay strategies of the robot are adaptively selected. The simulation results showed that the performance of the scheme is better than the existing scheme in terms of bit error rate, energy consumption, and efficiency.

In addition, accurate real-time adjustment of system adaptive parameters is also a condition for maintaining system stability and efficiency. For the CWI mentioned in [151–153], Abbasi et al. [163] used an adaptive filtering method. Its ANF needs to follow the interference, and it is a system that can find its coefficients according to the environment and the input signal. It uses a single neural network to predict filter coefficients, which is faster than traditional methods and improves the SNR significantly. Shi et al. [164] improved two algorithms, power distribution panel wavelet packet transform (PDP-WPT) and extended BP neural network (EBPNN), and used the adaptive TISI neural network to identify information in the transform domain. Compared with the traditional method, the capability of interference suppression can be improved by 32%.

3.5 Spectrum prediction

3.5.1 Definition & limitations

Spectrum prediction refers to the effective use of known spectrum data to predict the occupancy or idle situation in the spectrum at future moments [165]. It performs feature learning and summarizes experience to make reasonable predictions about the spectrum occupancy status of the future time slots.





Figure 14 (Color online) Spectrum prediction in covert satellite communication scenario.

Spectrum prediction is a complementary technique to spectrum sensing. A general spectrum prediction model is designed to provide accurate spectrum occupancy predictions for communication networks, which can allow SU to discover and access current spectrum holes. By use of spectrum prediction, the SE of the network will be maximized without interfering with primary user (PU) communications [166]. Spectrum prediction is also widely used in satellite communication, and can provide an effective solution for real-time spectrum resource allocation in large-scale satellite internet [167].

In covert satellite communication systems, the knowledge of the current spectrum is critically important. Because covert satellite communication is in an open space, the spectrum occupancy in the corresponding frequency band will greatly affect the reliability and covertness of the covert satellite communication system. By using spectrum prediction technology, users can flexibly select the frequency, bandwidth, and hopping spread spectrum mode of their own communication according to the current spectrum background, so as to actively bypass channels that may seriously interfere with communication performance. Signals that are currently occupied in the spectrum can also be used effectively as bunker signals. In this way, on the basis of ensuring the reliability of the communication, the covertness of the current spectrum state. However, for the reason that the communication delay between the satellite and the terrestrial station in the covert satellite communication system is much larger than that of the terrestrial communication system, the real-time performance of the cognition obtained spectrum has been seriously damaged [168]. Therefore, the satellite communication system needs to adopt spectrum prediction technology to predict the spectrum state in advance and improve the accuracy of perception. Figure 14 shows the spectrum prediction technique in the covert satellite communication scenario.

3.5.2 Main challenges and solutions

In the earliest days, most spectrum predictions were based on linear models, that was, by linearly changing and modeling the obtained historical spectrum data to obtain the future predicted values [26, 169–176]. For example, Huang et al. [171] proposed a spectrum prediction model based on partial periodic pattern mining, which improves the prediction performance under the conditions of sensing error and spectrum irregularity. Nguyen et al. [172] given a collaborative model based on spectrum prediction and spectrum sensing. In order to augment incomplete or corrupted historical observation data, Ding et al. [173] established a robust online spectrum prediction framework, and analyzed the impact of abnormal data on the rank distribution.

However, the methods above were limited to only solving linear problems. Most of the models in practice were nonlinear, and the predicted performance cannot be guaranteed in more complex scenarios. To adapt to the complex environments, Akbar et al. [177] proposed the use of the hidden Markov model (HMM) to model and predict spectrum occupancy. But, when the first-order hidden Markov model is applied to make predictions, the state of the current moment is only related to the previous moment, and the historical spectrum information is not fully utilized. Therefore, Chen et al. [178] proposed a higher-order hidden Markov model (HOHMM), which achieved better prediction performance. Noticing that cooperative spectrum prediction has the potential to improve prediction accuracy, Eltom et al. [179] proposed an HMM-based collaborative soft-fusion spectral prediction method, which can significantly reduce the prediction error.

However, the spectrum prediction method based on HMM also has some limitations. It needs to know the prior knowledge in the spectrum, and there are many parameters to be modified during learning and training, which results in large system loss.

3.5.3 AI-based solutions

In recent years, with the development of AI, more and more researchers apply AI to assist spectrum prediction techniques.

The neural network in DL is the first technique to be widely used in spectrum prediction. Zhang et al. [180] proposed a radial basis function (RBF) neural network spectrum prediction algorithm based on the K-means clustering algorithm (K-RBF). Lan et al. [181] designed a three-step advanced spectrum prediction framework based on a neural network. Mennes et al. [46] used a CNN to predict future spectrum usage. Zhang et al. [182] designed a multi-layer neural network predictor.

RNNs and their improved models such as LSTM have a strong ability to mine potential temporal correlations between input data. Yu et al. [53,54] first applied the LSTM network to spectrum prediction, the purpose of which was to better extract long-term features in time series spectrum data without using any prior information about the device. By learning from the past spectrum availability data, they exploited the inherent time-frequency correlation between the data to predict future spectrum availability. Applying the LSTM neural network for spectrum prediction, Yu et al. [55] effectively reduced the optimization time and consumed resources by using the Taguchi method instead of the grid search method. And in order to improve the timeliness of spectral prediction, they adopted the deep time spectral residual network (DTS-ResNet) [183]. By introducing advanced residual units, they constructed multiple residual network modules to capture the characteristics of these data at different time scales and predict multiple time series at the same time. For the problem of unreliable local spectrum prediction of a single user under harsh channel conditions, Shawel et al. [56] proposed a soft collaborative fusion model based on LSTM, which improves the prediction accuracy and optimizes the energy consumption.

However, the ability of only one kind of neural network was limited, so some researchers began to study multi-layer fusion neural networks. Yu et al. [184] designed a hierarchical spectrum learning system, and using fine-tuned CNN and gated recurrent unit network they proposed a DCG model for local spectral availability prediction. This hierarchical model combines the advantages of CNN and RNN at the same time, and has a strong ability to find correlations in the spatial and temporal domains of signals. Ding et al. [168] also proposed a CNN and bidirectional LSTM (CNN-BiLSTM) model, which was used on LEO to characterize the spectrum utilization law of geosynchronous earth orbit (GEO) satellites, and a fusion network was designed to effectively combine multiple prediction results.

The above-mentioned DL-based spectrum prediction studies all have an explicit or implicit assumption, that is, the historical data of the target frequency band are abundant. However, when the equipment for spectrum prediction switches to a new band or when the spectrum environment changes, traditional DL models will be limited by the timeliness and scarcity of required historical data. To overcome this issue, Davaslioglu et al. [68] used a GAN with a DL architecture to generate additional synthetic training data for spectral prediction. Lin et al. [185] proposed a cross-band spectrum prediction model based on transfer learning, and in the case of limited historical spectrum data, the performance was improved compared with the existing model. In order to further enhance the sparse spectral historical data, they proposed a cross-band spectral data enhancement framework DA-DTS-ResNet based on GAN and deep transfer learning [69]. Compared with the method in [68], this framework is simpler, more stable, easier to train, and can make the spectral prediction model more robust when data is scarce.

3.6 Adaptive modulation and coding

3.6.1 Definition & limitations

AMC is to get the current channel state by analyzing the information received by the receiver, and adaptively adjusting the modulation scheme and coding scheme of the sender at the next moment. AMC uses dynamically varying bit rate and modulation sequence to adapt to time-varying channels, thereby increasing capacity and reducing the bit error rate. With fixed code modulation, the system needs to be designed with link margin reserved for worst-case communication [186].

Covert satellite communication needs to work properly in the presence of a lot of interference. Therefore, the previous communication system would adopt the link allowance. By using low order modulation



Figure 15 (Color online) The system framework for AMC technology.

mode, low bit rate, and high transmit power, the system can still work normally under bad communication conditions. This method is relatively simple, but it wastes satellite power and spectrum resources. With the development of adaptive techniques, scholars began to consider the application of adaptive technology in satellite communication to overcome the above problems. AMC can compensate for the change of link-state caused by channel fading and environmental interference, and it has become a research hotspot in covert satellite communication. For the power-limited system such as covert satellite communication, AMC can effectively overcome the deficiency of power limitations. For a multi-beam system, different users can effectively match different link performances to maximize the overall throughput of the system. In covert satellite communication, the purpose of AMC is to dynamically adjust the coding rate and modulation mode to achieve higher data throughput and SE on the premise of ensuring the concealment of transmitted signals. In short, a high-order modulation mode and a high bit rate can be selected to improve the link throughput when the link condition is good. And in the case of poor link conditions, using low modulation mode and low bit rate can prevent the system from being affected.

Hayes [187] first proposed an adaptive receiver and feedback channel in 1968. In this method, the channel information obtained at the receiver is fed back to the transmitter, which adjusts the transmitted data through the feedback information. In 1992, Webb et al. [188] proposed a variable rate adaptive QAM modulation system. Simulation proved that under the same conditions, the channel SNR of the variable rate system is about 5 dB higher than that of the fixed 16-level QAM system. In 1999, considering that different levels of services are provided for different users in the cellular system, Qiu et al. [189] combined adaptive modulation and power control to improve the throughput of the whole communication system. In 2004, AMC was included in the DVB-S2 standard, which can increase the capacity gain by about 30% compared with the DVB-S standard [190]. In the same year, Rinaldo et al. [191] proposed an AMC scheme for multi-beam wideband satellite downlink, and they proved that compared with the traditional non-adaptive TDM system under the same conditions, the capacity of the proposed system increases by 2.5 times. In 2010, Ahn et al. [192] used irregularly modulated LDPC coding to reduce the system complexity of AMC and control the transmission rate. In 2021, based on the satellite DVB-S2/S2X standard, Tropea et al. [193] proposed a two-layer packet scheduling mechanism for AMC, which can support different QoS levels to monitor propagation and channel conditions, and the simulation results showed that the overall system performance is improved.

The system framework for AMC technology is shown in Figure 15. The receiver decodes the received information and estimates the transmission channel at the same time. Then the channel status information is fed back to the sender. According to the received channel status information, the sender determines the modulation and coding scheme (MCS) to be used for communication at the next moment, and sends the data according to the modulation and coding of the scheme.

3.6.2 Main challenges and solutions

In recent years, AMC has made remarkable progress in all aspects of research. According to the AMC technical structure, it can be roughly divided into the following directions. The first is to estimate the channel state. In the channel estimation of link adaptive technology, there are a lot of studies that use SNR as an indicator of channel evaluation. Traditional SNR estimation methods include maximum likelihood estimation, spectral estimation, statistics-based estimation, and so on [194]. It maps the current CSI to an SNR, which will be fed back to the sender for subsequent MCS switching. The second step is the selection and switch of the MCS scheme. The sender selects the encoding modulation scheme which is suitable for the current environment according to the MSC selection strategy.

Some advances in AMC technology in recent years are described as follows. The ultimate purpose of these studies is to optimize performance and improve spectrum utilization. The effectiveness of the AMC scheme depends on the accuracy of channel estimation. So channel estimation plays a key role in the AMC scheme. Weerackody [195] proposed a channel SNR estimation technique, in which the possible errors caused by channel estimation are considered and the performance of AMC is improved. Considering that rainfall and ground movement environment would cause serious attenuation to Ka-band signal communication, Yu et al. [196] derived a rain failure probability density function based on satellite elevation changes and calculated the channel parameters. Then they select the best MCS according to the adaptive coding algorithm and the simulation results showed that the proposed scheme can effectively compensate for attenuation. Wan et al. [197] proposed the effective signal-to-noise ratio (ESNR) as a new performance indicator for mode switching, and the simulation and experimental results verified the advantages of ESNR over other SNR definitions. Zeng et al. [198] have focused on fast-moving scenarios, which can enhance the time-domain equivalent wireless channel parameter correlation and reduce the channel quality indicator (CQI) mapping mismatch level in the communication environment with large feedback delay.

The switch and selection of the MCS scheme is also the key research. Meng et al. [199] studied the problem of constellation and channel coding rate selection in AMC for multiple-input multiple-output (MIMO) systems. They proposed a new selection rule for narrowband MIMO systems which selects constellation and coding rate by comparing throughput, and the rule was further promoted to broadband MIMO orthogonal frequency division multiplexing (OFDM) system. Farrokh et al. [200] expressed the MCS selection problem as a Markov decision problem approaching an infinite level, and the experiments proved that the optimal MCS selection strategy has a monotone structure under some sufficient conditions. Considering that channels fluctuate under the influence of shadow, multipath fading, and mobility, Kojima et al. [201] proposed AMC and frequency symbol extension (FSS) adaptive switching strategies for OFDM systems to improve throughput performance by using a mixture of FSS and traditional AMC. There are also some other scholars focusing on the improvement of channel coding [192] and channel prediction [202].

3.6.3 AI-based solutions

With the development of AI, more and more scholars began to consider the combination of ML and AMC to improve throughput, spectrum efficiency, and the accuracy of CSI estimation while reducing computational complexity. Elwekeil et al. [203] compared the performance of K-nearest Neighbor, SVM, and DL when they were respectively applied to AMC, and analyzed the system throughput and algorithm complexity of the three schemes. Ferreira et al. [204] integrated multi-objective RL and ANN to achieve the management of available resources and task-based conflict objectives, which can be used in the flexible and changeable satellite communication environment. The simulation results proved that the algorithm can improve system capacity.

Some scholars optimize AMC by improving the accuracy of channel estimation. AbdelMoniem et al. [205] proposed an efficient channel estimation algorithm based on LSTM neural network. Compared with the traditional non-orthogonal multiple access system, this channel estimation algorithm reduces the interrupt probability by 10% on average. Kojima et al. [206] combined an ANN with SNR estimation and established the mapping relationship between the power spectral density value and different MCS through training. They improved the system throughput and reduced the computational complexity by 97.5% compared with using the traditional EVM-based AMC scheme. Tato et al. [207] selected modulation schemes using a multilayer feedforward neural network-assisted with SNR estimation.

There are also many research achievements in the switching and selection of MCS. Zhang et al. [208] utilized DRL and proposed an intelligent MCS selection algorithm for master transmission. The simulation results showed that without considering the cost, the primary transmission rate of the proposed algorithm is 90%-100% of the optimal MCS selection scheme. Subsequently, they applied supervised learning and RL to AMC technology respectively [209] to meet the contradictory requirements of high data rate and high reliability by adjusting modulation sequence and coding rate. Zhang et al. [43] used a CNN to identify modulation formats, which avoids artificial feature selection and provides good classification accuracy. In addition, AI has been combined with research on channel decoding [210] and CSI [211] prediction.

The key to AMC is the assumption that real-time channel parameter estimates can be received. But channel changes rapidly in satellite communication, which poses a challenge to the accuracy of AMC. AMC based on ML is a novel direction, and the neural network algorithm shows great advantages in prediction. Therefore, with the continuous breakthrough of AI algorithms, it is believed that there will be more algorithms with excellent performance combined with traditional AMC techniques.

4 Future trends

In this section, future development trends will be discussed based on the above research and discussion about key techniques in covert satellite communication.

(1) LPD communication. At present, the research on LPD communication mainly focuses on LPD information theory, implementation scheme, constant rate covert communication, and so on. However, AI is seldom used in LPD communication. The future development trend of LPD communication can be considered as follows. (a) Generate more featureless signals with good performance through ML and optimize the LPD model by using optimization algorithms. (b) Consider key generation in communication through training networks. At present, there is a lot of theoretical analysis on key length and usage, but there is still a lack of practical application on how to generate keys using actual channel characteristics. (c) The eavesdropper and communication parties present a game relationship and consider the combination of game theory and LPD communication. Especially in the multi-user communication network, multiple communication parties can jointly fight against the monitor and get higher system throughput.

(2) Transmit power control. Recently, the transmit power control technique in covert satellite communication mainly focuses on four problems: improving network throughput, enhancing network robustness, reducing control delay, and enhancing covertness. AI applications are limited to solving the first three problems to improve the reliability of covert satellite communication. Research on the power control of AI to improve covertness is yet to be carried out. The use of satellite-ground networks will increase in the future, and in this scenario, it will be an important topic to consider the reliability and covertness of network node communication. AI will play an irreplaceable role.

(3) Phased array antenna. At present, AI algorithms have achieved good results in beamforming and antenna calibration. In terms of calibration accuracy and transmitting beam performance, the AI algorithm continues to surpass traditional algorithms. However, there is still some research on hardware design, and the antenna is mainly designed by manual work. How to use AI to design the antenna system with the best performance according to the application scenarios and environmental parameters is a major research direction in the future. In addition, new application directions such as AI fault detection also need to be more developed and mined to build a complete intelligent phased array antenna system.

(4) Anti-interference. The intelligence of interference detection and anti-interference decision has become a developing trend. However, most existing research related to AI has high implementation complexity and is not suitable to complete relevant operations on the satellite. Algorithm optimization and complexity reduction are important to research direction in the future. In addition, most of the communication channels considered at present are Gaussian white noise channels, and non-human interference such as adjacent satellites and solar wind is hardly considered. In the future, more research is needed on interference identification and anti-interference decision in complex environments.

(5) Spectrum prediction. Spectrum prediction has been widely used, which greatly improves spectrum utilization. Future research on spectrum prediction will more concentrate on large-scale MIMO. How to use multi-node cooperative prediction in AI coordination communication networks to improve prediction accuracy is an important research direction. On the other hand, using AI data enhancement to make faster predictions with fewer data is also a promising direction.

(6) Adaptive modulation and coding. In covert satellite communication systems, AMC techniques based on AI have gradually matured. But current AMC techniques based on ML mostly use a single AI algorithm. One of the future research trends is based on combinatorial ML. In addition, the performance of the AMC technique is poor in high-speed channel environments. It is a hot issue that needs further study in the future to use AI to learn and classify scenarios with rapid channel changes. AI can also be used to achieve more accurate channel state prediction and higher robustness.

In the future, with the continuous deepening of the urbanization process around the world, there will be more and more demands for covert satellite communication in complex urban environments. The covert satellite communication scene in the urban environment is shown in Figure 16. In the urban environment, the space is filled with signals of various forms and various frequency bands. This complex electromagnetic environment is both an opportunity and a challenge for covert satellite communication.



Figure 16 (Color online) The covert satellite communication scene in the urban environment.

How to use the complex electromagnetic environment to improve the covertness of communication while keeping the reliability unaffected by it will be an important topic in the future. At the same time, in the city, hidden users will face the threat of close detection, that is to say, eavesdroppers may exist in any corner of the city, and may be very close to the covert users, which will lead to a very high SNR at the Eavesdroppers. In that case, the traditional methods of hiding communication signals below the noise floor are no longer realistic, and new covert communication mechanisms must be studied.

5 Conclusion

In this study, a comprehensive overview of the application of AI in covert satellite communication systems was presented. This study summarized the background and applications of covert satellite communication and clarified the importance of the six key techniques in covert satellite communication systems, including LPD communication, transmit power control, phased array antenna, anti-interference, spectrum prediction, and AMC. In particular, the research status of these techniques and the application of AI in them were introduced in detail. Finally, the existing problems and future research directions of these techniques were pointed out. Obviously, covert satellite communication will be an important part of future satellite communication, and it has broad prospects in both commercial and military fields.

Acknowledgements This work was supported by National Natural Science Foundation of China (Grant Nos. 44562101050, 62001022, U1836201, 61971038).

References

- 1 You X H, Wang C-X, Huang J, et al. Towards 6G wireless communication networks: vision, enabling technologies, and new paradigm shifts. Sci China Inf Sci, 2021, 64: 110301
- 2 Liu D K, Zhang Z S, Ke S, et al. Joint angle delay estimation in terahertz large-scale array system. Sci China Inf Sci, 2020, 63: 180304
- 3 Wang C H, Zhang Z S, Wu J Y, et al. An overview of protected satellite communications in intelligent age. Sci China Inf Sci, 2021, 64: 161301
- 4 Ke S, An J P, Wang S. Synthesis-free directional modulation for retrodirective frequency diverse array. Sci China Inf Sci, 2020, 63: 202304
- 5 Yue P, An J, Zhang J, et al. On the security of LEO satellite communication systems: Vulnerabilities, countermeasures, and future trends. 2022. arXiv.2201.03063
- 6 Petitcolas F A P, Anderson R J, Kuhn M G. Information hiding—a survey. Proc IEEE, 1999, 87: 1062–1078
- 7 Chen C, Wang S, Li L, et al. Intelligent covert satellite communication for military robot swarm. IEEE Access, 2019, 8: 5363–5382
- 8 Xie A-H, Zhu L-D, Zhai J-Q, et al. A method of designing covert DSSS-signal for anti-blind detection (in Chinese). Acta Electron Sin, 2018, 46: 2817–2823

- 9 Wang S, Chen S, Wang A, et al. Joint timing and channel estimation for bandlimited long-code-based MC-DS-CDMA: a low-complexity near-optimal algorithm and the CRLB. IEEE Trans Commun, 2013, 61: 1998–2011
- 10 Liu Y, Ning P, Dai H, et al. Randomized differential DSSS: jamming-resistant wireless broadcast communication. In: Proceedings of the 29th Conference on Information Communications, San Diego, 2010. 695–703
- 11 Chung J H, Yang K. Optimal frequency-hopping sequences with new parameters. IEEE Trans Inform Theor, 2010, 56: 1685–1693
- 12 Luo S X, Zhang Z S, Wang S, et al. Network for hypersonic UCAV swarms. Sci China Inf Sci, 2020, 63: 140311
- 13 Gao L, Zhang Y, Pan S. Research on power control strategy in satellite overlap communication system (in Chinese). Radio Commun Tech, 2016, 42: 21–23, 27
- 14 Xue R, Wang Y. A satellite overlapping covert communication scheme based on LDPC-DCSK transmission technology. In: Proceedings of the 9th International Conference on Networks, Communication and Computing, Tokyo, 2020
- 15 Zhang B, Zhang J, Guo D. Overlay communication based on satellite with DSSS signal (in Chinese). J Commun, 2005, 26: 57–62
- 16 Koivisto T, Koivunen V. Blind despreading of short-code DS-CDMA signals in asynchronous multi-user systems. Signal Processing, 2007, 87: 2560-2568
- 17 Tsatsanis M K, Giannakis G B. Blind estimation of direct sequence spread spectrum signals in multipath. IEEE Trans Signal Process, 1997, 45: 1241–1252
- 18 Gu X, Zhao Z, Shen L. Blind estimation of pseudo-random codes in periodic long code direct sequence spread spectrum signals. IET Commun, 2016, 10: 1273–1281
- 19 Wyner A D. The wire-tap channel. Bell Syst Technical J, 1975, 54: 1355–1387
- 20 Yan S, Zhou X, Hu J, et al. Low probability of detection communication: opportunities and challenges. IEEE Wireless Commun Mag, 2019, 26: 19-25
- 21 Chai X M, Xu X, Zhang Z S. A user-selected uplink power control algorithm in the two-tier femtocell network. Sci China Inf Sci, 2015, 58: 042303
- 22 Lu K, Cong Y, Fang J, et al. A joint code-doppler acquisition algorithm for DSSS-MSK based on FFT. In: Proceedings of IEEE 4th Information Technology, Networking, Electronic and Automation Control Conference (ITNEC), Chongqing, 2020. 1118–1123
- 23 Parker D, Zimmermann D C. Phased arrays-part 1: theory and architectures. IEEE Trans Microwave Theor Techn, 2002, 50: 678-687
- 24 Parker D, Zimmermann D C. Phased arrays-part II: implementations, applications, and future trends. IEEE Trans Microwave Theor Techn, 2002, 50: 688–698
- 25 Gao G X, Sgammini M, Lu M, et al. Protecting GNSS receivers from jamming and interference. Proc IEEE, 2016, 104: 1327–1338
- 26 Acharya P A K, Singh S, Zheng H. Reliable open spectrum communications through proactive spectrum access. In: Proceedings of the 1st International Workshop on Technology and Policy for Accessing Spectrum, Association for Computing Machinery, New York, 2006
- 27 Fan W F, Lu P, Tian C. Unifying logic rules and machine learning for entity enhancing. Sci China Inf Sci, 2020, 63: 172001
 28 Mi H B, Xu K L, Feng D W, et al. Collaborative deep learning across multiple data centers. Sci China Inf Sci, 2020, 63: 182102
- 29 Ning Z L, Sun S M, Wang X J, et al. Intelligent resource allocation in mobile blockchain for privacy and security transactions: a deep reinforcement learning based approach. Sci China Inf Sci, 2021, 64: 162303
- 30 Ye N, An J, Yu J. Deep-learning-enhanced NOMA transceiver design for massive MTC: challenges, state of the art, and future directions. IEEE Wireless Commun, 2021, 28: 66–73
- 31 Ye N, Li X, Yu H, et al. Deep learning aided grant-free NOMA toward reliable low-latency access in tactile Internet of Things. IEEE Trans Ind Inf, 2019, 15: 2995–3005
- 32 Chan K S, Zary N. Applications and challenges of implementing artificial intelligence in medical education: integrative review. JMIR Med Educ, 2019, 5: e13930
- 33 Lin M, Zhao Y. Artificial intelligence-empowered resource management for future wireless communications: a survey. China Commun, 2020, 17: 58–77
- 34 Simeone O. A very brief introduction to machine learning with applications to communication systems. IEEE Trans Cogn Commun Netw, 2018, 4: 648–664
- 35 Wang J, Jiang C, Zhang H, et al. Thirty years of machine learning: the road to pareto-optimal wireless networks. IEEE Commun Surv Tutorials, 2020, 22: 1472–1514
- 36 Du J, Jiang C, Wang J, et al. Machine learning for 6G wireless networks: carrying forward enhanced bandwidth, massive access, and ultrareliable/low-latency service. IEEE Veh Technol Mag, 2020, 15: 122–134
- 37 Fourati F, Alouini M-S. Artificial intelligence for satellite communication: a review. Intell Converged Networks, 2021, 2: 213–243
- 38 Huang G, Ge C J, Xiong T Y, et al. Large scale air pollution prediction with deep convolutional networks. Sci China Inf Sci, 2021, 64: 192107
- 39 Li W X, Lin N, Zhang M Z, et al. VNet: a versatile network to train real-time semantic segmentation models on a single GPU. Sci China Inf Sci, 2022, 65: 139105
- 40 Ferre R M, de la Fuente A, Lohan E S. Jammer classification in GNSS bands via machine learning algorithms. Sensors, 2019, 19: 4841
- 41 Wu Z, Zhao Y, Yin Z, et al. Jamming signals classification using convolutional neural network. In: Proceedings of IEEE International Symposium on Signal Processing and Information Technology (ISSPIT), Bilbao, 2017. 62-67
- 42 Kartchner D R, Palmer R, Jayaweera S K. Satellite navigation anti-spoofing using deep learning on a receiver network. In: Proceedings of IEEE Cognitive Communications for Aerospace Applications Workshop (CCAAW), Cleveland, 2021. 1–5
- 43 Zhang T, Wang D, Wang D, et al. Modulation format recognition based on CNN in satellite communication system. In: Proceedings of IEEE 21st International Conference on Communication Technology (ICCT), 2021. 1392–1396
 44 Jiang J, Wang Z, Zhao H, et al. Modulation recognition method of satellite communication based on cldnn model.
- In: Proceedings of IEEE 30th International Symposium on Industrial Electronics (ISIE), Kyoto, 2021. 1–6
- 45 Elbir A M, Mishra K V. Joint antenna selection and hybrid beamformer design using unquantized and quantized deep learning networks. IEEE Trans Wireless Commun, 2020, 19: 1677–1688
- 46 Mennes R, Claeys M, de Figueiredo F A P, et al. Deep learning-based spectrum prediction collision avoidance for hybrid

wireless environments. IEEE Access, 2019, 7: 45818–45830

- 47 Lee W, Kim M, Cho D H. Deep power control: transmit power control scheme based on convolutional neural network. IEEE Commun Lett, 2018, 22: 1276–1279
- 48 Lee K, Lee J R, Choi H H. Learning-based joint optimization of transmit power and harvesting time in wireless-powered networks with co-channel interference. IEEE Trans Veh Technol, 2020, 69: 3500–3504
- 49 Gherrity M. A learning algorithm for analog, fully recurrent neural networks. In: Proceedings of International 1989 Joint Conference on Neural Networks, Washington, 1989. 643–644
- 50 Hochreiter S, Schmidhuber J. Long short-term memory. Neural Computation, 1997, 9: 1735–1780
- 51 Gunn L, Smet P, Arbon E, et al. Anomaly detection in satellite communications systems using LSTM networks. In: Proceedings of Military Communications and Information Systems Conference (MilCIS), 2018
- 52 Lee S, Kim S, Seo M, et al. Synchronization of frequency hopping by LSTM network for satellite communication system. IEEE Commun Lett, 2019, 23: 2054–2058
- 53 Yu L, Chen J, Ding G. Spectrum prediction via long short term memory. In: Proceedings of the 3rd IEEE International Conference on Computer and Communications (ICCC), Chengdu, 2017. 643–647
- 54 Yu L, Wang Q, Guo Y, et al. Spectrum availability prediction in cognitive aerospace communications: a deep learning perspective. In: Proceedings of Cognitive Communications for Aerospace Applications Workshop (CCAA), 2017. 1–4
- 55 Yu L, Chen J, Ding G, et al. Spectrum prediction based on taguchi method in deep learning with long short-term memory. IEEE Access, 2018, 6: 45923–45933
- 56 Shawel B S, Woledegebre D H, Pollin S. Deep-learning based cooperative spectrum prediction for cognitive networks. In: Proceedings of International Conference on Information and Communication Technology Convergence (ICTC), Jeju, 2018. 133-137
- 57 Tsuchida H, Kawamoto Y, Kato N, et al. Efficient power control for satellite-borne batteries using Q-learning in low-earthorbit satellite constellations. IEEE Wireless Commun Lett, 2020, 9: 809–812
- 58 Xiong D N, Yi L. Q-learning based handoff algorithm for satellite system with ancillary terrestrial component. J Commun, 2015, 36: 252
- 59 Yao F, Jia L, Sun Y, et al. A hierarchical learning approach to anti-jamming channel selection strategies. Wireless Netw, 2019, 25: 201–213
- 60 Noori H, Vilni S. Jamming and anti-jamming in interference channels: a stochastic game approach. IET Commun, 2020, 14: 682–692
- 61 Han C, Liu A, Wang H, et al. Dynamic anti-jamming coalition for satellite-enabled army IoT: a distributed game approach. IEEE Internet Things J, 2020, 7: 10932–10944
- 62 Nasir Y S, Guo D. Multi-agent deep reinforcement learning for dynamic power allocation in wireless networks. IEEE J Sel Areas Commun, 2019, 37: 2239–2250
- 63 Wang M Q, Yuan W W, Zhang J Y. Overview of research on generative adversarial network GAN (in Chinese). Comput Eng Design, 2021, 42: 3389–3395
- 64 Goodfellow I, Pouget-Abadie J, Mirza M, et al. Generative adversarial nets. In: Proceedings of Advances in Neural Information Processing Systems, 2014
- 65 Moti Z, Hashemi S, Namavar A. Discovering future malware variants by generating new malware samples using generative adversarial network. In: Proceedings of the 9th International Conference on Computer and Knowledge Engineering (ICCKE), Mashhad, 2019. 319–324
- 66 Creswell A, White T, Dumoulin V, et al. Generative adversarial networks: an overview. IEEE Signal Process Mag, 2018, 35: 53–65
- 67 Guan Q, Shuang W. Detection for multisatellite downlink signal based on generative adversarial neural network. Math Problems Eng, 2020, 2020: 1–14
- 68 Davaslioglu K, Sagduyu Y E. Generative adversarial learning for spectrum sensing. In: Proceedings of 2018 IEEE International Conference on Communications (ICC), Kansas City, 2018. 1–6
- 69 Lin F, Chen J, Ding G, et al. Spectrum prediction based on GAN and deep transfer learning: a cross-band data augmentation framework. China Commun, 2021, 18: 18–32
- 70 Liao X, Si J, Shi J, et al. Generative adversarial network assisted power allocation for cooperative cognitive covert communication system. IEEE Commun Lett, 2020, 24: 1463–1467
- 71 Fischer C M, Wartick M, Mark M M. Detection probability and taxpayer compliance: a review of the literature. J Accounting Literature, 1992, 11: 1–46
- 72 Ding X, An J, Zhao Z, et al. Low-density parity-check coded direct sequence spread spectrum receiver based on analog probabilistic processing. IEEE Trans Veh Technol, 2021, 70: 6355–6370
- 73 Pickholtz R, Schilling D, Milstein L. Theory of spread-spectrum communications—a tutorial. IEEE Trans Commun, 1982, 30: 855–884
- 74 Scholtz R. The spread spectrum concept. IEEE Trans Commun, 1977, 25: 748-755
- 75 Bark G. Power control and active channel selection in an LPI FH system for HF communications. In: Proceedings of Military Communications Conference, 1997. 1031–1035
- 76 Scholtz R. The origins of spread-spectrum communications. IEEE Trans Commun, 1982, 30: 822-854
- 77 Cooper G R, Nettleton R W. A spread-spectrum technique for high-capacity mobile communications. IEEE Trans Veh Technol, 1978, 27: 264–275
- 78 Xing Y, Hu A, Zhang J, et al. On radio frequency fingerprint identification for DSSS systems in low SNR scenarios. IEEE Commun Lett, 2018, 22: 2326–2329
- 79 Liu F, Marcellin M W, Goodman N A, et al. Compressive sampling for detection of frequency-hopping spread spectrum signals. IEEE Trans Signal Process, 2016, 64: 5513–5524
- 80 Bash B A, Goeckel D, Towsley D. Limits of reliable communication with low probability of detection on AWGN channels. IEEE J Sel Areas Commun, 2013, 31: 1921–1930
- 81 Abdelaziz A, Koksal C E. Fundamental limits of covert communication over MIMO AWGN channel. In: Proceedings of IEEE Conference on Communications and Network Security (CNS), Las Vegas, 2017. 1–9
- 82 Wang L, Wornell G W, Zheng L. Fundamental limits of communication with low probability of detection. IEEE Trans Inform Theor, 2016, 62: 3493–3503
- 83 Tahmasbi M, Bloch M R. First- and second-order asymptotics in covert communication. IEEE Trans Inform Theor, 2019, 65: 2190–2212

- 84 Bloch M R. Covert communication over noisy channels: a resolvability perspective. IEEE Trans Inform Theor, 2016, 62: 2334–2354
- 85 Che P H, Bakshi M, Jaggi S. Reliable deniable communication: hiding messages in noise. In: Proceedings of IEEE International Symposium on Information Theory, Istanbul, 2013. 2945–2949
- 86 Bloch M R, Guha S. Optimal covert communications using pulse-position modulation. In: Proceedings of IEEE International Symposium on Information Theory (ISIT), Aachen, 2017. 2825–2829
- 87 Kadampot I A, Tahmasbi M, Bloch M R. Multilevel-coded pulse-position modulation for covert communications. In: Proceedings of IEEE International Symposium on Information Theory (ISIT), Vail, 2018. 1864–1868
- 88 Li R, Cui J, Huang T, et al. Optimal pulse-position modulation order and transmit power in covert communications. IEEE Trans Veh Technol, 2022, 71: 5570–5575
- 89 He B, Yan S, Zhou X, et al. On covert communication with noise uncertainty. IEEE Commun Lett, 2017, 21: 941–944
- 90 Ta H Q, Kim S W. Covert communication under channel uncertainty and noise uncertainty. In: Proceedings of IEEE International Conference on Communications (ICC), Shanghai, 2019. 1–6
- 91 Bash B A, Goeckel D, Towsley D. Covert communication gains from adversary's ignorance of transmission time. IEEE Trans Wireless Commun, 2016, 15: 8394–8405
- 92 Sobers T V, Bash B A, Guha S, et al. Covert communication in the presence of an uninformed jammer. IEEE Trans Wireless Commun, 2017, 16: 6193–6206
- 93 Sobers T V, Bash B A, Goeckel D, et al. Covert communication with the help of an uninformed jammer achieves positive rate. In: Proceedings of the 49th Asilomar Conference on Signals, Systems and Computers, Pacific Grove, 2015. 625–629
- 94 Shahzad K, Zhou X, Yan S, et al. Achieving covert wireless communications using a full-duplex receiver. IEEE Trans Wireless Commun, 2018, 17: 8517–8530
- 95 Yang L, Yang W, Xu S, et al. Achieving covert wireless communications using a full-duplex multi-antenna receiver. In: Proceedings of IEEE 5th International Conference on Computer and Communications (ICCC), Chengdu, 2019. 912–916
- 96 He B, Yan S, Zhou X, et al. Covert wireless communication with a poisson field of interferers. IEEE Trans Wireless Commun, 2018, 17: 6005–6017
- 97 Yang Y, Zhou J, Wang F, et al. An LPI design for secure burst communication systems. In: Proceedings of IEEE China Summit International Conference on Signal and Information Processing (ChinaSIP), Xi'an, 2014. 631–635
- 98 Shakeel I. Machine learning based featureless signalling. In: Proceedings of IEEE Military Communications Conference (MILCOM), Los Angeles, 2018. 1–9
- 99 Zhou J, Zheng S, Yu X, et al. Low probability of intercept communication based on structured radio beams using machine learning. IEEE Access, 2019, 7: 169946
- 100 Sun J, Zhang Z, Xiao H, et al. Uplink interference coordination management with power control for D2D underlaying cellular networks: modeling, algorithms, and analysis. IEEE Trans Veh Technol, 2018, 67: 8582–8594
- 101 Du C, Zhang Z, Wang X, et al. Optimal duplex mode selection for D2D-aided underlaying cellular networks. IEEE Trans Veh Technol, 2020, 69: 3119-3134
- 102 Du C, Zhang Z, Wang X, et al. Deep learning based power allocation for workload driven full-duplex D2D-aided underlaying networks. IEEE Trans Veh Technol, 2020, 69: 15880–15892
- 103 Khan S, Shin S Y. Deep learning aided transmit power estimation in mobile communication system. IEEE Commun Lett, 2019, 23: 1405–1408
- 104 Liu C, Zhang Z. On connectivity of flying ad hoc networks in the presence of ground terminal. China Commun, 2021, 18: 1-10
- 105 Hu J, Yan S, Zhou X, et al. Covert communications without channel state information at receiver in IoT systems. IEEE Internet Things J, 2020, 7: 11103–11114
- 106 Wang J, Guo D, Zhang B, et al. Spectrum access and power control for cognitive satellite communications: a game-theoretical learning approach. IEEE Access, 2019, 7: 164216
- 107 Dvorkind T G. Power allocation for covert communication, with application to underwater acoustic channel. In: Proceedings of IEEE International Conference on the Science of Electrical Engineering (ICSEE), Eilat, 2016. 1–5
- 108 Shannon C E. A mathematical theory of communication. Bell Syst Technical J, 1948, 27: 379–423
- Ioo Zuo P, Peng T, Linghu W, et al. Optimal resource allocation for hybrid interweave-underlay cognitive SatCom uplink.
 In: Proceedings of IEEE Wireless Communications and Networking Conference (WCNC), Barcelona, 2018. 1–6
- 110 Gao B, Lin M, An K, et al. ADMM-based optimal power control for cognitive satellite terrestrial uplink networks. IEEE Access, 2018, 6: 64757–64765
- 111 Vassaki S, Poulakis M I, Panagopoulos A D, et al. Power allocation in cognitive satellite terrestrial networks with QoS constraints. IEEE Commun Lett, 2013, 17: 1344–1347
- 112 Shi S, An K, Li G, et al. Optimal power control in cognitive satellite terrestrial networks with imperfect channel state information. IEEE Wireless Commun Lett, 2018, 7: 34–37
- 113 Hu J, Li G, Bian D, et al. Optimal power control for cognitive LEO constellation with terrestrial networks. IEEE Commun Lett, 2020, 24: 622–625
- 114 Shi S, Li G, An K, et al. Optimal power control for real-time applications in cognitive satellite terrestrial networks. IEEE Commun Lett, 2017, 21: 1815–1818
- 115 Wang W, Zhao S, Zheng Y, et al. Resource allocation method of cognitive satellite terrestrial networks under non-ideal spectrum sensing. IEEE Access, 2019, 7: 7957–7964
- 116 Letzepis N. Optimal power allocation for parallel gaussian channels with LPD constraints. In: Proceedings of IEEE Military Communications Conference, Baltimore, 2016. 103–107
- 117 Forouzesh M, Azmi P, Kuhestani A. Secure transmission with covert requirement in untrusted relaying networks. In: Proceedings of the 9th International Symposium on Telecommunications (IST), Tehran, 2018. 670–675
- 118 Hu J, Yan S, Zhou X, et al. Covert communication achieved by a greedy relay in wireless networks. IEEE Trans Wireless Commun, 2018, 17: 4766–4779
- 119 Zheng T X, Wang H M, Ng D W K, et al. Multi-antenna covert communications in random wireless networks. IEEE Trans Wireless Commun, 2019, 18: 1974–1987
- 120 Hu J, Yan S, Zhou X, et al. Covert wireless communications with channel inversion power control in Rayleigh fading. IEEE Trans Veh Technol, 2019, 68: 12135–12149
- 121 Ma R, Yang X, Pan G, et al. Covert communications with channel inversion power control in the finite blocklength regime. IEEE Wireless Commun Lett, 2021, 10: 835–839

- 122 Wang M, Yang W, Lu X, et al. Channel inversion power control aided covert communications in uplink NOMA systems. IEEE Wireless Commun Lett, 2022, 11: 871–875
- 123 Hornik K, Stinchcombe M, White H. Multilayer feedforward networks are universal approximators. Neural Networks, 1989, 2: 359–366
- 124 Sun H, Chen X, Shi Q, et al. Learning to optimize: training deep neural networks for interference management. IEEE Trans Signal Process, 2018, 66: 5438–5453
- 125 Lee W. Resource allocation for multi-channel underlay cognitive radio network based on deep neural network. IEEE Commun Lett, 2018, 22: 1942–1945
- 126 Lee W, Jo O, Kim M. Intelligent resource allocation in wireless communications systems. IEEE Commun Mag, 2020, 58: 100–105
- 127 Lee W, Lee K. Resource allocation scheme for guarantee of QoS in D2D communications using deep neural network. IEEE Commun Lett, 2021, 25: 887–891
- 128 Lee W, Lee K. Deep learning-aided distributed transmit power control for underlay cognitive radio network. IEEE Trans Veh Technol, 2021, 70: 3990–3994
- 129 Zhang H, Yang N, Huangfu W, et al. Power control based on deep reinforcement learning for spectrum sharing. IEEE Trans Wireless Commun, 2020, 19: 4209–4219
- 130 Li X, Fang J, Cheng W, et al. Intelligent power control for spectrum sharing in cognitive radios: a deep reinforcement learning approach. IEEE Access, 2018, 6: 25463-25473
- 131 Tan J, Liang Y C, Zhang L, et al. Deep reinforcement learning for joint channel selection and power control in D2D networks. IEEE Trans Wireless Commun, 2021, 20: 1363–1378
- 132 Fan Z, Hong B, Du H. Review on anti-jamming technology for phased-array antenna of satellite. Space Electronic Tech, 2016, 13: 61–66
- 133 Meng M, Shi Y, Jiang Z, et al. A review of the development of space phased array antenna technology. In: Proceedings of the 3rd Aerospace Electronics Strategy Research Forum, Beijing, 2017
- 134 Liu N. Research on key technologies of phased array antenna of vehicle satellite communication. Dissertation for Ph.D. Degree. Beijing: Beijing Institute of Technology, 2015
- 135 Lin M, Gong Z. Smart antenna and its application in satellite communication anti-jamming (in Chinese). Telecommun Sci, 2003, 19: 40-44
- 136 Kapusuz K Y, Sen Y, Bulut M, et al. Low-profile scalable phased array antenna at Ku-band for mobile satellite communications. In: Proceedings of IEEE International Symposium on Phased Array Systems and Technology (PAST), Waltham, 2016. 1–4
- 137 Rao J B L, Mital R, Patel D P, et al. Low-cost phased array antenna for satellite communications on mobile earth stations.
 In: Proceedings of IEEE International Symposium on Phased Array Systems and Technology, Waltham, 2013. 214–219
- 138 Zhang H T, Wang W, Jin M P, et al. An active phased array antenna for broadband mobile satellite communications at ka-band. In: Proceedings of CIE International Conference on Radar (RADAR), Guangzhou, 2016. 1–3
- 139 Kumar B P, Kumar C, Kumar V S, et al. A spherical phased array antenna with unequal amplitude excitation for satellite application. In: Proceedings of IEEE International Conference on Antenna Innovations Modern Technologies for Ground, Aircraft and Satellite Applications (iAIM), Bangalore, 2017. 1–4
- 140 Moon S M, Yun S, Yom I B, et al. Phased array shaped-beam satellite antenna with boosted-beam control. IEEE Trans Antennas Propagat, 2019, 67: 7633–7636
- 141 Raj K V, Ranjitha S, Meghana V, et al. Satellite tracking using 7×7 hexagonal phased array antenna. In: Proceedings of the 4th International Conference on Recent Trends on Electronics, Information, Communication Technology (RTEICT), Bangalore, 2019. 369–374
- 142 Lin Y, Ma Q, Wang S, et al. Calibration for spaceborne phased array antennas without interrupting satellite communications. In: Proceedings of the 9th International Conference on Wireless Communications and Signal Processing (WCSP), Nanjing, 2017. 1–5
- 143 Ghaffarian N, Wahab W A, Raeesi A, et al. Characterization and calibration challenges of an k-band large-scale active phased-array antenna with a modular architecture. In: Proceedings of the 50th European Microwave Conference (EuMC), Utrecht, 2021. 1039–1042
- 144 Mishchenko S E, Shatskij V V, Eliseev D Y, et al. Neural network approach to the solution of constructive synthesis problems of active phased antenna arrays. In: Proceedings of Radiation and Scattering of Electromagnetic Waves (RSEMW), Divnomorskoe, 2019. 256–260
- 145 Lovato R, Gong X. Phased antenna array beamforming using convolutional neural networks. In: Proceedings of IEEE International Symposium on Antennas and Propagation and USNC-URSI Radio Science Meeting, Atlanta, 2019. 1247–1248
- 146 Fournier J L, Titz D, Ferrero F, et al. Phased array antenna controlled by neural network FPGA. In: Proceedings of Loughborough Antennas Propagation Conference, Loughborough, 2011. 1–5
- 147 Hameed I, Tuan P V, Koo I. Deep learning-based energy beamforming with transmit power control in wireless powered communication networks. IEEE Access, 2021, 9: 142795
- 148 Sarayloo Z, Masoumi N, Shahi H, et al. A convolutional neural network approach for phased array calibration using poweronly measurements. In: Proceedings of the 28th Iranian Conference on Electrical Engineering (ICEE), Tabriz, 2020. 1–6
- 149 Southall H L, Simmers J A, O'Donnell T H. Direction finding in phased arrays with a neural network beamformer. IEEE Trans Antennas Propagat, 1995, 43: 1369–1374
- 150 Savasta S, Presti L L, Rao M. Interference mitigation in GNSS receivers by a time-frequency approach. IEEE Trans Aerosp Electron Syst, 2013, 49: 415–438
- 151 Lv Q, Qin H. A novel algorithm for adaptive notch filter to detect and mitigate the CWI for GNSS receivers. In: Proceedings of IEEE 3rd International Conference on Signal and Image Processing (ICSIP), Shenzhen, 2018. 444–451
- 152 Chien Y R. Design of GPS anti-jamming systems using adaptive notch filters. IEEE Syst J, 2015, 9: 451–460
- 153 Mao W L, Ma W J, Chien Y R, et al. New adaptive all-pass based notch filter for narrowband/FM anti-jamming GPS receivers. Circ Syst Signal Process, 2011, 30: 527–542
- 154 Wang Z, Lv M, Tang B. Paper application of partial coefficient update LMS algorithm to suppress narrowband interference in DSSS system. In: Proceedings of International Conference on Communication Software and Networks, Chengdu, 2009. 275–278
- 155 Zhang L, Yuan S, Chen Y, et al. Narrowband interference suppression in DSSS system based on frequency shift wavelet packet transform. In: Proceedings of the 9th International Conference on Electronic Measurement Instruments, Beijing,

2009. 333-337

- 156 Djukanovic S, Dakovic M, Stankovic L. Local polynomial fourier transform receiver for nonstationary interference excision in DSSS communications. IEEE Trans Signal Process, 2008, 56: 1627–1636
- 157 Amin M G, Wang X, Zhang Y D, et al. Sparse arrays and sampling for interference mitigation and DOA estimation in GNSS. Proc IEEE, 2016, 104: 1302–1317
- 158 Heng L, Walter T, Enge P, et al. GNSS multipath and jamming mitigation using high-mask-angle antennas and multiple constellations. IEEE Trans Intell Transp Syst, 2014, 16: 741–750
- 159 Zhao H, Shi Y, Zhang B, et al. Analysis and simulation of interference suppression for space-time adaptive processing. In: Proceedings of IEEE International Conference on Signal Processing, Communications and Computing (ICSPCC), Guilin, 2014. 724–727
- 160 Xu H, Cheng Y, Liang J, et al. A jamming recognition algorithm based on deep neural network in satellite navigation system. In: Proceedings of China Satellite Navigation Conference, Chengdu, 2020. 701–711
- 161 Erpek T, Sagduyu Y E, Shi Y. Deep learning for launching and mitigating wireless jamming attacks. IEEE Trans Cogn Commun Netw, 2019, 5: 2–14
- 162 Lu X Z, Jie J F, Lin Z H, et al. Reinforcement learning based energy efficient robot relay for unmanned aerial vehicles against smart jamming. Sci China Inf Sci, 2022, 65: 112304
- 163 Abbasi M, Mosavi M R, Reazei M J. GPS continues wave jamming canceller using an ANF combined with an artificial neural network. In: Proceedings of the 8th Iranian Joint Congress on Fuzzy and intelligent Systems (CFIS), Mashhad, 2020. 99–104
- 164 Shi Y, Liu Y. Interference suppression for satellite navigation based on neural network and wavelet packets transform. Adv Materials Res, 2012, 457–458: 1111–1117
- 165 Ding G, Jiao Y, Wang J, et al. Spectrum inference in cognitive radio networks: algorithms and applications. IEEE Commun Surv Tutorials, 2018, 20: 150–182
- 166 Xing X, Jing T, Cheng W, et al. Spectrum prediction in cognitive radio networks. IEEE Wireless Commun, 2013, 20: 90–96
- 167 Zhang Y, Wu Y, Liu A, et al. Deep learning-based channel prediction for LEO satellite massive MIMO communication system. IEEE Wireless Commun Lett, 2021, 10: 1835–1839
- 168 Ding X, Feng L, Zou Y, et al. Deep learning aided spectrum prediction for satellite communication systems. IEEE Trans Veh Technol, 2020, 69: 16314–16319
- 169 Makhoul J. Linear prediction: a tutorial review. Proc IEEE, 1975, 63: 561–580
- 170 Wen Z, Fan C, Zhang X, et al. A learning spectrum hole prediction model for cognitive radio systems. In: Proceedings of the 10th IEEE International Conference on Computer and Information Technology, 2010. 2089–2093
- 171 Huang P, Liu C J, Yang X, et al. Wireless spectrum occupancy prediction based on partial periodic pattern mining. IEEE Trans Parallel Distrib Syst, 2014, 25: 1925–1934
- 172 Nguyen V D, Shin O S. Cooperative prediction-and-sensing-based spectrum sharing in cognitive radio networks. IEEE Trans Cogn Commun Netw, 2018, 4: 108–120
- 173 Ding G, Wu F, Wu Q, et al. Robust online spectrum prediction with incomplete and corrupted historical observations. IEEE Trans Veh Technol, 2017, 66: 8022–8036
- 174 Chen X, Yang J, Ding G. Minimum Bayesian risk based robust spectrum prediction in the presence of sensing errors. IEEE Access, 2018, 6: 29611–29625
- 175 Chen X, Yang J. A spectrum prediction-based frequency band pre-selection over deteriorating HF electromagnetic environment. China Commun, 2018, 15: 10–24
- 176 Wen Z, Luo T, Xiang W, et al. Autoregressive spectrum hole prediction model for cognitive radio systems. In: Proceedings of IEEE International Conference on Communications Workshops, Beijing, 2008. 154–157
- 177 Akbar I A, Tranter W H. Dynamic spectrum allocation in cognitive radio using hidden markov models: Poisson distributed case. In: Proceedings of IEEE SoutheastCon, Richmond, 2007. 196–201
- 178 Chen Z, Qiu R C. Prediction of channel state for cognitive radio using higher-order hidden Markov model. In: Proceedings of the IEEE SoutheastCon 2010, Concord, 2010. 276–282
- 179 Eltom H, Kandeepan S, Liang Y C, et al. Cooperative soft fusion for HMM-based spectrum occupancy prediction. IEEE Commun Lett, 2018, 22: 2144–2147
- 180 Zhang S, Hu J, Bao Z, et al. Prediction of spectrum based on improved RBF neural network in cognitive radio. In: Proceedings of International Conference on Wireless Information Networks and Systems (WINSYS), Reykjavik, 2013. 1–5
- 181 Lan K, Zhao H, Zhang J, et al. A spectrum prediction approach based on neural networks optimized by genetic algorithm in cognitive radio networks. In: Proceedings of the 10th International Conference on Wireless Communications, Networking and Mobile Computing (WiCOM 2014), Beijing, 2014. 131–136
- 182 Zhang Y, Hou J, Towhidlou V, et al. A neural network prediction-based adaptive mode selection scheme in full-duplex cognitive networks. IEEE Trans Cogn Commun Netw, 2019, 5: 540–553
- 183 Yu L, Chen J, Zhang Y, et al. Deep spectrum prediction in high frequency communication based on temporal-spectral residual network. China Commun, 2018, 15: 25–34
- 184 Yu L, Guo Y, Wang Q, et al. Spectrum availability prediction for cognitive radio communications: a DCG approach. IEEE Trans Cogn Commun Netw, 2020, 6: 476–485
- 185 Lin F, Chen J, Sun J, et al. Cross-band spectrum prediction based on deep transfer learning. China Commun, 2020, 17: 66–80
- 186 Bischl H, Brandt H, de Cola T, et al. Adaptive coding and modulation for satellite broadband networks: from theory to practice. Int J Satell Commun Network, 2010, 28: 59–111
- 187 Hayes J. Adaptive feedback communications. IEEE Trans Commun, 1968, 16: 29–34
- 188 Webb W T, Steele R. Variable rate QAM for mobile radio. IEEE Trans Commun, 1995, 43: 2223–2230
- 189 Qiu X X, Chawla K. On the performance of adaptive modulation in cellular systems. IEEE Trans Commun, 1999, 47: 884–895
- 190 Morello A, Mignone V. DVB-S2: the second generation standard for satellite broad-band services. Proc IEEE, 2006, 94: 210–227
- 191 Rinaldo R, Gaudenzi R D. Capacity analysis and system optimization for the forward link of multi-beam satellite broadband systems exploiting adaptive coding and modulation. Int J Satell Commun Network, 2004, 22: 401–423
- 192 Ahn S K, Yang K. Adaptive modulation and coding schemes based on LDPC codes with irregular modulation. IEEE Trans

Commun, 2010, 58: 2465-2470

- 193 Tropea M, de Rango F, Santamaria A F. Design of a two-stage scheduling scheme for DVB-S2/S2X satellite architecture. IEEE Trans Broadcast, 2021, 67: 424–437
- 194 Pauluzzi D R, Beaulieu N C. A comparison of SNR estimation techniques for the AWGN channel. IEEE Trans Commun, 2000, 48: 1681–1691
- 195 Weerackody V. Adaptive coding and modulation for satellite communication links in the presence of channel estimation errors. In: Proceedings of IEEE Military Communications Conference, Diego, 2013. 622–627
- Yu X, Wang S. Adaptive information transmission scheme for LEO satellite based on Ka band. Comput Sci, 2019, 46: 72–79
 Wan L, Zhou H, Xu X, et al. Adaptive modulation and coding for underwater acoustic OFDM. IEEE J Ocean Eng, 2015,
- 40: 327–336198 Zeng R, Liu T, Yu X T, et al. Novel channel quality indicator prediction scheme for adaptive modulation and coding in high
- mobility environments. IEEE Access, 2019, 7: 11543–11553 199 Meng J, Yang E H. Constellation and rate selection in adaptive modulation and coding based on finite blocklength analysis
- and its application to LTE. IEEE Trans Wireless Commun, 2014, 13: 5496–5508
- 200 Farrokh A, Krishnamurthy V, Schober R. Optimal adaptive modulation and coding with switching costs. IEEE Trans Commun, 2009, 57: 697–706
- 201 Kojima S, Maruta K, Ahn C J. Throughput maximization by adaptive switching with modulation coding scheme and frequency symbol spreading. JCOMSS, 2018, 14: 332–339
- 202 Zheng Y, Ren S, Xu X, et al. A modified ARIMA model for CQI prediction in LTE-based mobile satellite communications. In: Proceedings of IEEE International Conference on Information Science and Technology, Wuhan, 2012. 822-826
- 203 Elwekeil M, Jiang S, Wang T, et al. Deep convolutional neural networks for link adaptations in MIMO-OFDM wireless systems. IEEE Wireless Commun Lett, 2019, 8: 665–668
- 204 Ferreira P V R, Paffenroth R, Wyglinski A M, et al. Multiobjective reinforcement learning for cognitive satellite communications using deep neural network ensembles. IEEE J Sel Areas Commun, 2018, 36: 1030–1041
- 205 AbdelMoniem M, Gasser S M, El-Mahallawy M S, et al. Enhanced NOMA system using adaptive coding and modulation based on LSTM neural network channel estimation. Appl Sci, 2019, 9: 3022
- 206 Kojima S, Maruta K, Ahn C J. Adaptive modulation and coding using neural network based SNR estimation. IEEE Access, 2019, 7: 183545–183553
- 207 Tato A, Mosquera C, Henarejos P, et al. Neural network aided computation of mutual information for adaptation of spatial modulation. IEEE Trans Commun, 2020, 68: 2809–2822
- 208 Zhang L, Tan J, Liang Y C, et al. Deep reinforcement learning-based modulation and coding scheme selection in cognitive heterogeneous networks. IEEE Trans Wireless Commun, 2019, 18: 3281–3294
- 209 Zhang L, Wu Z. Machine Learning—Based Adaptive Modulation and Coding Design. Piscataway: Wiley-IEEE Press, 2020. 157–180
- 210 Gruber T, Cammerer S, Hoydis J, et al. On deep learning-based channel decoding. In: Proceedings of the 51st Annual Conference on Information Sciences and Systems (CISS), Baltimore, 2017. 1–6
- 211 Wang X, Li H, Wu Q. Optimizing adaptive coding and modulation for satellite network with ML-based CSI prediction. In: Proceedings of IEEE Wireless Communications and Networking Conference (WCNC), Shenzhen, 2019. 1–6

Appendix A

Acronyms	Definition	Acronyms	Definition
ADMM	Alternating direction method of multipliers	HMM	Hidden Markov model
AI	Artificial intelligence	HOHMM	Higher-order hidden Markov model
AMC	Adaptive modulation and coding	IF	Intermediate frequency
ANF	Adaptive notch filter	IIR	Infinite impulse response
ANFA	All-pass based notch filter	IoT	Internet of things
ANN	Artificial neural network	LEO	Low earth orbit
APAA	Active phased array antenna	LPD	Low probability of detection
AWGN	Additive white Gaussian noise	LPI	Low probability of intercept
BSC	Binary symmetric channel	LSTM	Long short-term memory
CIPC	Channel inversion power control	MCS	Modulation and coding scheme
CNN	Convolutional neural network	MIMO	Multiple-input multiple-output
CRN	Cognitive radio network	ML	Machine learning
CSI	Channel state information	OFDM	Orthogonal frequency division multiplexing
CWI	Continuous wave interference	PPM	Pulse position modulation
DL	Deep learning	PU	Primary user
DMC	Discrete memoryless channel	QoS	Quality of service
DNN	Deep neural network	RBF	Radial basis function
DPC	Deep power control	RL	Reinforcement learning
DRL	Deep reinforcement learning	RNN	Recurrent neural network
DSSS	Direct sequence spread spectrum	RSS	Received signal strength
EE	Energy efficiency	SE	Spectral efficiency
EIRP	Equivalent isotropic radiated power	SNR	Signal-to-noise ratio
ESNR	Effective signal-to-noise ratio	SRL	Square root law
FHSS	Frequency hopping spread spectrum	SS	Spread spectrum
FIR	Finite impulse response	STAP	Space-time adaptive processing
FSS	Frequency symbol extension	SU	Secondary user
GAN	Generative adversarial network	SVM	Support vector machine
GEO	Geosynchronous earth orbit	WMMSE	Weighted minimum mean square error

 Table A1
 List of acronyms